#### ARTHUR ZIMMERMANN

# Tumors and Tumor-Like Lesions of the Hepatobiliary Tract

General and Surgical Pathology



## Tumors and Tumor-Like Lesions of the Hepatobiliary Tract

#### Arthur Zimmermann

### Tumors and Tumor-Like Lesions of the Hepatobiliary Tract

General and Surgical Pathology

With 880 Figures and 199 Tables



Arthur Zimmermann Emeritus Professor of Pathology University of Bern Bern, Switzerland

ISBN 978-3-319-26954-2 ISBN 978-3-319-26956-6 (eBook) ISBN 978-3-319-26955-9 (print and electronic bundle) DOI 10.1007/978-3-319-26956-6

Library of Congress Control Number: 2016948216

#### © Springer International Publishing Switzerland 2017

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made.

Printed on acid-free paper

This Springer imprint is published by Springer Nature
The registered company is Springer International Publishing AG
The registered company address is: Gewerbestrasse 11, 6330 Cham, Switzerland

I dedicate this textbook to my wife, Geneviève, who for many years of preparatory work endured much and offered continued interest, help, and compassion while I was creating this work. She accompanied me with great loyalty in this endeavor. I also dedicate this book to my children, Laxmi and Tristan, who have given so much meaning to my life, and to my venerated late teacher in pathology, Professor Hans Cottier.

#### **Preface**

This textbook is designed to be a comprehensive assessment of current knowledge regarding the surgical and general pathology of hepatobiliary tumors and tumor-like lesions. The scope of the book is broad and provides an up-to-date source for the wide-ranging tumor pathology of the entire hepatobiliary tract. In the planning phase of this work, the question had arisen as to the purpose and need of such a book, in the light of numerous excellent monographs and textbooks that have been published in this field during the last years, but a major justification for a new book relates to the rapid change in the role of pathology in the investigation of hepatobiliary tumors and related lesions. Therefore, an update of the dramatic changes that have taken place in the discovery and application of several lines of knowledge referring to hepatobiliary tumor pathology in a broader sense was regarded a worthwhile task. Notwithstanding the impressive increase in diagnostic precision of modern imaging and other, in particular various molecular, methods, tumor biopsy and its morphological interpretation based on complex techniques is still a central diagnostic instrument that serves refined diagnosis and classification, risk stratification, and therapy planning, also in the light of future personalized treatment strategies. A combined approach by using conventional, fine structural, immunohistochemical and hybridization morphological studies, and molecular techniques generated a new concept of the tight correlation between structure and function in tumor pathology, contributing to advanced modes of diagnosis. Apart from information on a given diagnostic tumor entity, differential diagnosis is discussed in depth as a most critical issue.

The main reason for integrating several important issues of general pathology is based on the rapidly evolving and continued changes that are occurring in the disciplines of tumor biology, genomics, and associated molecular features that characterize tumors. The concept of the present work in fact aims at concentrating detailed aspects of surgical pathology needed for diagnosis and the pathogenic mechanisms behind disease in one source with ample color illustrations and a detailed reference corpus.

In the light of refined imaging techniques and other modern diagnostic approaches that can uncover a host of previously undetectable hepatobiliary lesions, a significant part of the textbook is dedicated to an extensive range of tumor-like lesions, pathologies that may, in a clinical-radiological setting, be confounded with true hepatobiliary neoplasms. Part of the tumor-like lesions, including mass-forming infections and infestations, are common entities

viii Preface

world-wide, while others may appear as unexpected or incidental findings, or rare and "exotic" disorders.

This text has been planned to serve hepatopathologists, hepatologists, and others interested and involved in this field, and it is the author's hope that the book is a comprehensive account on the surgical and general pathology of hepatobiliary tumors and their tumor-like mimics. The book is also hoped to be a useful source of information for basic scientists active in the field of liver pathophysiology.

To provide a systematic review of the immense field of hepatobiliary tumors and tumor-like lesions, the textbook has been divided into 39 parts, each covering one to several chapters, in order to assist the reader in locating topics of interest. In what follows, brief overviews on the contents of each chapter are presented.

Part 1 starts off with a series of chapters that supply comprehensive information on tumors characterized by a hepatocyte-derived lineage and its precursors. Chapter ▶ 1 deals with the role of hepatic stem and progenitor cells in hepatocarcinogenic pathways. Hepatocytes were perceived to represent the major cells of origin for numerous neoplasms of the liver, but stem and progenitor cells have been identified as important sources. The chapter addresses issues of hepatic stem cell niches, types of stem/progenitor cells found in these niches, interactions of stem cells with other cells, changes of their microenvironment, and mechanisms involved in a stem cell-cancer sequence. As cancer initiating cells and cancer-associated stem cells can circulate in blood, the significance of cycling clonogenic cells with longevity and remote spread for tumor progression is discussed. Chapter ▶ 2 provides an in-depth description of ordinary hepatocellular carcinoma (HCC), including classification of gross phenotypes, macroscopic growth patterns, pertinent histologic and diagnostic features, and tumor grading. In Chap. ▶ 3, the numerous immunohistochemical features characterizing ordinary HCC are discussed in detail. Chapter ▶ 4 focuses on invasion and metastatic patterns of ordinary HCC. Patterns of macrovascular and microvascular invasion and the features of intrahepatic and remote metastasis are explained and illustrated. This part also provides information on risk factors for metastasis and on the presentation and frequency of extrahepatic organ metastases. In Chap. ▶ 5, secondary changes that develop in HCCs, and in particular the interesting phenomenon of spontaneous tumor regression, are highlighted. Progression and recurrence of HCC are major elements of the tumor's biology of disease. Numerous prognostic factors for the natural course of HCC have been delineated, discussed in detail in Chap. ▶ 6. The issues of Chap. ▶ 7 are the various types of HCC precursor lesions that can develop in cirrhotic livers, including small and large cell change, dysplastic foci, and dysplastic nodules. There is a group of ordinary HCCs characterized by small size at the time point of diagnosis, lesions that are more frequently diagnosed due to improved imaging techniques. These intriguing lesions, specifically their morphology, classification, biology, and relationship to early cancer are dealt with in Chap.  $\triangleright$  8. The chapters on neoplasms of the hepatocyte lineage are considerably extended to reflect the growing importance of special types of liver cell cancers in the setting of clinical presentation, detectability by modern imaging techniques,

Preface ix

molecular features, and biology of disease (Chaps.  $\triangleright$  9,  $\triangleright$  10,  $\triangleright$  11,  $\triangleright$  12, and ▶ 13). Chapter ▶ 9 deals with clear cell HCC, a group of neoplasms that belong to a growing spectrum of epithelial clear cell tumors of the alimentary tract with a distinct biology of disease. A heterogeneous group of HCCs is characterized by the accumulation of neutral fat, including steatotic HCC and its inflammatory variant, steatohepatitic HCC, tumors that also develop in the setting of nonalcoholic and alcoholic fatty liver disease. Part of these tumors are rich in Mallory-Denk bodies (Chap. ▶ 10). A rare group of HCCs is characterized by the presence of an abundant desmoplastic stroma, similar to cholangiocarcinoma (sclerosing and scirrhous HCCs; Chap. ▶ 11). A further unusual subset of HCCs shows dense infiltrates of mononuclear leukocytes (inflammatory HCCs). One variant of these neoplasms reflects a morphology similar to lymphoepithelial carcinoma, with or without association with EBV virus infection, and another rare variant exhibits a plasmacellular infiltrate and signs of regression (medullary HCC; Chap. ▶ 12). Very rare forms of HCCs are characterized by peliotic change, multinucleated giant cells, chromophobe cells, oxyophilic/oncocytic cells, or cells with a Dubin-Johnson-like pigment (Chap. ≥ 13). An interesting group of liver cell tumors displays the presence of progenitor cells or stem-like cells. Part of these HCCs with progenitor cell features express cytokeratin 19, a feature conferring a more aggressive course (Chap. ▶ 14). A clinically and radiologically intriguing situation is produced by HCCs arising in an ectopic, extrahepatic location (Chap. ▶ 15). HCCs also occur in infancy and childhood (pediatric HCC). It is not yet clarified whether these unusual malignancies are the same or different from their adult counterparts (Chap. ▶ 16). A neoplasm that in several respects mimics HCC is hepatoid carcinoma, which can develop in numerous organs, but is usually manifest in the liver in the form of metastases (Chap. ▶ 17). An intriguing variant of HCC is fibrolamellar HCC, a tumor mainly occurring in younger individuals, showing a biology similar to that of ordinary HCC, and associated with a typical recurrent chimeric transcript (Chap. ▶ 18). A rare group of neoplasms composed of immature hepatocyte progenitors (embryonal and fetal hepatocytes) is formed by the various types and subtypes of hepatoblastoma and related neoplasms (Chap. ▶ 19). The majority of these cancers develops in children younger than 5 years, but rarely also develop in adults. Unusual variants of hepatoblastoma with aberrant differentiation patterns are treated in Chap. ▶ 20, including tumors with a bile duct cell differentiation. The focus of Chap. ▶ 21 relates to the biology and prognostic factors of heptoblastoma, while Chap. ▶ 22 treats risk factors and pathogenic pathways of neoplasms of the hepatoblastoma tumor family. An intriguing neoplasm related to hepatoblastoma and associated with an interesting clinical presentation and unique molecular features, nested stromal epithelial tumor, is the theme of Chap.  $\triangleright$  23. A benign tumor of the hepatocyte lineage is hepatocellular adenoma, which has recently been subdivided into several molecular subtypes associated with distinct morphologic patterns. The discussion of this important hepatic tumor and its variants is found in Chaps. > 24 and  $\triangleright$  25. A rare group of neoplasms containing hepatocyte-like cells is combined hepatocellular-cholangiocarcinoma, malignancies that display a biphenotypic histologic picture. In contrast to HCC, these highly aggressive

x Preface

neoplasms occur in cirrhotic and noncirrhotic livers with almost the same frequency (Chap. ▶ 26).

Part 2 of the textbook relates to benign and malignant neoplasms of the cholangiocyte lineage. Cholangiocarcinomas (CC) are divided into two major groups, extrahepatic and intrahepatic bile duct cancers. Among the former, hilar and perihilar CC form a distinct clinicopathologic entity different from CC originating in the mid-region and distal parts of the large bile duct. Intrahepatic CC is a malignancy that can present in three major gross growth patterns and originates from both small or large intrahepatic bile ducts. CC are malignancies of adult patients, but very rarely also develop in the pediatric age group (Chaps.  $\triangleright$  27,  $\triangleright$  28,  $\triangleright$  29,  $\triangleright$  30,  $\triangleright$  31, and  $\triangleright$  32). A distinct group of bile duct neoplasms is formed by intraductal neoplasms, tumors that resemble their pancreatic counterparts. They display a phase of intraluminal, often papillary noninvasive growth and a later high risk of transition into invasive CC (Chap. ≥ 33). In addition to classical forms of CC, there are tumors developing in the setting of hepatobiliary cystic disease, or exhibit distinct differentiation patterns different from those observed in ordinary CC (Chaps.  $\triangleright$  34,  $\triangleright$  35,  $\triangleright$  36,  $\triangleright$  37, and  $\triangleright$  38). A rare subset of bile duct tumors is characterized by cysts lined by a mucin-producing epithelium, frequently associated with a subepithelial ovarian-like stroma and expression of sex steroid receptors. These mucinous cystic neoplasms (MCN) can undergo a dysplasia-carcinoma sequence (Chap. ≥ 39). Both the intrahepatic and extrahepatic biliary tree can be the site of several types of benign epithelial neoplasms and hamartomas, including tubular and papillary adenomas, peribiliary gland hamartomas, biliary microhamartoma, and neoplasms and hyperplasias of peribiliary glands (Chaps.  $\triangleright$  40 and  $\triangleright$  41).

Part 3 covers a heterogeneous group of liver tumors derived from other epithelial lineages. Some hepatobiliary carcinomas are characterized by varying proportions of squamous epithelial cells, including squamous cell carcinoma, adenosquamous carcinoma, and mucoepidermoid carcinoma, or acinar cell and adenoid cystic components (Chaps. ▶ 42 and ▶ 43). Rare carcinomas of the biliary tract are undifferentiated neoplasms, such as nonendocrine small cell and spindle cell carcinomas, and carcinomas with rhabdoid features (Chap. ▶ 44). In a small group of hepatobiliary neoplasms, the cells of origin are not yet fully clarified. These lesions mainly comprise hepatic adrenal rest tumor and progenitor cell neoplasms (Chap. ▶ 45).

Part 4 of the textbook refers to mixed epithelial-mesenchymal tumors of the hepatobiliary tract. Chapter ▶ 46 highlights an intriguing group of hepatobiliary malignancies that are composed of a complex mixture of various neoplastic tissue types. These tumors are classified as carcinosarcomas and carcinomas with sarcomatoid features. Few primary hepatic tumors are characterized by the presence of multinucleated, osteoclast-like giant cells with a macrophage/histiocyte lineage (Chap. ▶ 47). Apart from carcinosarcomas, rare hepatic tumors present a mixed cellular phenotype that is still difficult to classify, including malignant mixed tumors, adenosarcomas, stromal tumors, and adult-type mixed hepatoblastomas (Chap. ▶ 48).

Part 5 focuses on the predominant group of hepatobiliary mesenchymal tumors, i.e., vascular tumors. The most important type of primary hepatic

Preface xi

vascular neoplasm is cavernous hemangioma with its various phenotypes and associations with extrahepatic vascular tumors (Chap.  $\triangleright$  49). Less common hepatic vascular tumors include several forms of hemangiomatosis, hemangioblastoma, epithelioid hemangioendothelioma, infantile hepatic hemangioma/hemangioendothelioma, kaposiform hemangioendothelioma, angiosarcoma, Kaposi's sarcoma, glomus tumors, myopericytoma, and glomangiopericytoma (Chaps.  $\triangleright$  50,  $\triangleright$  51,  $\triangleright$  52,  $\triangleright$  53,  $\triangleright$  54,  $\triangleright$  55,  $\triangleright$  56, and  $\triangleright$  57). Unusual vascular malformations, which may be part of complex inborn syndromes, can mimic true vascular neoplasms, such as Klippel-Trenaunay syndrome (Chap.  $\triangleright$  58). A second group of reactive vascular lesions that can cause tumor-like hepatic manifestations include bacillary angiomatosis and peliosis hepatis (Chap.  $\triangleright$  59).

Part 6 refers to tumors and tumor-like lesions of lymph vessels. In the hepatobiliary tract, these are very rare conditions which include cystic and noncystic lymphangioma, capillary lymphangioma, lymphangiomatosis, hepatic lymphangiectasis, and lymphocele (Chap. ▶ 60).

Part 7 is exclusively dedicated to solitary fibrous tumor and tumors with a hemangiopericytoma-like pattern. These lesions now include at least part of the former hemangiopericytomas and are characterized by a distinct somatic fusion of two genes, NAB2 and STAT6 (Chap.  $\triangleright$  61).

Part 8 discusses the complex spectrum of nonvascular mesenchymal tumors of the hepatobiliary tract. All these neoplasm are rare lesions and include fibroblastoid and myofibroblastoid neoplasms, leiomyoma and leiomyosarcomas, rhabdomyosarcomas, lipoma, liposarcoma, myelolipoma, hibernoma, tumors with osteosarcomatous and chondrosarcomatous components, gastrointestinal stromal tumors, benign and malignant nerve sheath tumors, granular cell tumors, synovial sarcoma, and undifferentiated high-grade pleomorphic sarcomas (Chaps.  $\triangleright$  62,  $\triangleright$  63,  $\triangleright$  64,  $\triangleright$  65,  $\triangleright$  66,  $\triangleright$  67,  $\triangleright$  68,  $\triangleright$  69,  $\triangleright$  70, and  $\triangleright$  71).

Part 9 summarizes tumors with a mesothelial cell lineage. Primary mesotheliomas of the liver are very rare, but characteristic neoplasms mimic mesothelial tumors in other locations. A unique mesothelial tumor occurring also in the liver is adenomatoid tumor (Chap.  $\triangleright$  72).

The theme of Part 10 is a heterogeneous group of neoplasms that are derived from, or related to, perivascular epithelioid cells. The liver is the primary site of various perivascular epithelioid cell tumors or PEComas, all with complex cellular compositions. They include PEComa proper, angiomyolipoma with its various subtypes, clear cell myomelanocytic tumors, clear cell and sugar tumors, and lymphangiomyomatosis (Chap. ▶ 73).

Part 11 presents hepatobiliary tumors of neuroendocrine lineages. Chapter ▶ 74 provides pertinent information regarding extraadrenal paraganglioma, neoplasms that also in the liver range in biology from a benign to frankly malignant behavior. In Chap. ▶ 75, the various types and subtypes of hepatobiliary neuroendocrine tumors are treated, emphasis being placed on novel classifications and grading systems.

In Part 12, a rare group of small tumors that also originate in the liver are discussed. The lesions comprise various small cell blue tumors, such as primitive neuroectodermal tumors (PNET), desmoplastic small round cell tumor, NUT midline carcinoma, and hepatic neuroblastoma (Chap. ▶ 76).

xii Preface

Part 13 refers to the interesting group of primary and secondary melanotic tumors of the hepatobiliary tract. Emphasis is placed on primary and metastatic melanoma, in particular also metastatic ocular melanoma, melanotic progonoma, and manifestations of melanoma of soft parts (Chap. > 77).

Part 14 focuses on hepatic tumors with a rhabdoid cell lineage. Liver and bile duct tumors with rhabdoid cell components are, at least in part, associated with absence of the chromatin remodeling factor SRF5/INI1 and include malignant rhabdoid tumor proper, carcinomas with rhabdoid features, and a subset of small cell hepatoblastoma (Chap. ▶ 78).

In Part 15, primary and metastatic germ cell tumors are treated. Most germ cell tumors occurring in the gonads can occur as primary lesions in the liver, but teratomas, yolk sac tumor, and choriocarcinoma prevail, also in the pediatric age group. The liver is a well-known site of metastatic germ cell tumors and can be the site of growing teratoma syndrome (Chap. ▶ 79).

Part 16 summarizes the pathology of hepatic manifestations of myeloid neoplasms. With the exception of granulocytic sarcomas, these neoplasms cause diffuse infiltration of the liver substance. The conditions discussed comprise polycythemia vera, several types of myeloproliferative syndrome, chronic eosinophilic leukemia and idiopathic hypereosinophilia, mast cell neoplasms, acute leukemias, myeloid neoplasms with a monocytoid lineage, and blastic plasmacytoid dendritic cell neoplasms (Chaps.  $\triangleright$  80,  $\triangleright$  81,  $\triangleright$  82,  $\triangleright$  83,  $\triangleright$  84,  $\triangleright$  85,  $\triangleright$  86, and  $\triangleright$  87).

Part 17 refers to the complex pathology of hepatobiliary Hodgkin's disease. This disorder causes, on the one hand, tumorous hepatic lesions that can clinically be confounded with liver cancer, but on the other hand also reveals associations with paraneoplastic changes, including vanishing bile duct syndrome (Chap. ▶ 88).

Part 18 covers the large field of hepatobiliary non-Hodgkin's lymphomas, other lymphoproliferative disorders, and neoplasms of dendritic and histiocytic cell systems. Major groups comprise B-cell and T-cell neoplasms that occur in numerous extrahepatic sites, but pseudolymphomas, neoplasms of the Langerhans cell and histiocytic systems, dendritic cell neoplasms, and reactive histiocytic syndromes (such as Rosai-Dorfman syndrome) are also discussed (Chaps.  $\triangleright$  89,  $\triangleright$  90,  $\triangleright$  91,  $\triangleright$  92,  $\triangleright$  93,  $\triangleright$  94,  $\triangleright$  95,  $\triangleright$  96,  $\triangleright$  97,  $\triangleright$  98,  $\triangleright$  99,  $\triangleright$  100,  $\triangleright$  101,  $\triangleright$  102, and  $\triangleright$  103).

Part 19 treats mesenchymal hamartoma of the liver and related neoplasms. Mesenchymal hamartoma and undifferentiated embryonal sarcoma of the liver are typical pediatric hepatic neoplasms, but they have rare counterparts in adult patients (Chaps.  $\triangleright$  104 and  $\triangleright$  105).

Part 20 is a large complex of chapters that relates to the very important issue of metastatic liver disease. The chapters discuss in detail aspects of gross and microscopic pathology of liver metastases in general, a specific chapter on colorectal cancer metastases, other common and rare metastatic cancers, secondary changes that frequently develop in hepatic metastases, secondary spread of metastatic disease into locoregional lymph nodes, associated liver lesions, growth and regrowth of metastases, and pathogenic features of liver metastasis (Chaps. ▶ 106, ▶ 107, ▶ 108, ▶ 109, ▶ 110, ▶ 111, ▶ 112, ▶ 113, ▶ 114, and ▶ 115).

Preface xiii

In Part 21, a small theme of liver tumor pathology is addressed, tumors and tumor-like lesions of hepatic ligaments. Falciform and round hepatic ligaments are the site of rare primary benign and malignant neoplasms, metastases, and various types of cysts (Chap.  $\triangleright$  116).

Part 22 contains chapters on reactive nodular hyperplastic hepatocyte lesions of the liver. Focal nodular hyperplasia (FNH) of the liver is an important mass-forming regenerative condition that often develops secondary to localized vascular and circulatory abnormalities of the liver. After hemangiomas, FNH is the second most common benign hepatic tumorous lesion (Chap. ▶ 117). A second important regenerative condition of the liver is nodular regenerative hyperplasia, which is associated with a broad array of causative factors (Chap. ▶ 118).

In Part 23, pseudotumors of the hepatobiliary tract are discussed. Pseudotumors and inflammatory pseudotumors form a heterogeneous group of lesions that share spindle cell proliferations and inflammatory infiltrates of variable density. Inflammatory myofibroblastic tumors is a lesion that contains a subset with neoplastic features and aberrant ALK expression (Chap. ▶ 119).

Part 24 relates to nonneoplastic tumor-like lesions of the liver. The liver is the site of tumor-like ectopias and heterotopias, mass-forming malformations, solitary necrotic nodule, various types of dust-induced nodular lesions, tumor-like lesions caused by gallstones and foreign bodies, pseudotumors consisting of reactive proliferations of hematopoietic cells and macrophages, pyogenic liver abscesses mimicking cancer, numerous hepatic bacterial, fungal, and protozoal infections causing tumor-like hepatic masses (tuberculosis, syphilis, brucellosis, and amebiasis representing prominent examples), tumor-like parasitic lesions (mainly echinococcosis), and liver infarcts (Chaps. ▶ 120, ▶ 121, ▶ 122, ▶ 123, ▶ 124, ▶ 125, ▶ 126, ▶ 127, ▶ 128, ▶ 129, ▶ 130, ▶ 131, ▶ 132, ▶ 133, ▶ 134, ▶ 135, ▶ 136, and ▶ 137).

Part 25 refers to reactive cystic lesions of the liver that may mimic cystic neoplasms. They include simple nonparasitic cysts, ciliated foregut cyst, pancreatitic pseudocysts, and cerebrospinal fluid pseudocysts (Chap. ▶ 138).

Part 26 provides information related to hepatic mass lesions caused by noninfectious granulomas. The main disorder is sarcoidosis that can also cause a complex form of sclerosing bile duct disease with bile duct loss. Blau syndrome and complex inflammatory disorders in part involving deregulated inflammasome function are also discussed (Chap. ▶ 139). A small chapter refers to chronic granulomatous disease (Chap. ▶ 140).

Part 27 presents the hepatic pathology of interesting fibrosclerotic disorders. The conditions include idiopathic retroperitoneal fibrosis and its variants, and the complex spectrum of IgG4-associated systemic sclerosing disease (Chap. ▶ 141).

In Part 28, numerous reactive bile duct alterations that can mimic biliary neoplasms are discussed. Intrahepatic and extrahepatic bile ducts can be involved with inflammatory stenosing polyps, granulomatous cholangitis, follicular cholangitis, oriental cholangitis, xanthogranulomatous cholangitis, bile duct cholesterolosis, sclerosing eosinophilic cholangitis, mechanical and anatomical bile duct alterations, bile duct stenosis caused by congenital

xiv Preface

anomalies and acquired disorders of the splanchnic arterial tree and the portal vein, and postcholecystectomy changes (Chaps. ▶ 142, ▶ 143, ▶ 144, ▶ 145, and ▶ 146).

Part 29 addresses the important issue of gallbladder cancer and other tumors and tumor-like lesions of this organ. Ordinary gallbladder carcinoma usually develops in a gallbladder that has undergone secondary changes related to longstanding cholelithiasis and associated inflammations. This neoplasm, an adenocarcinoma, can be associated with epithelial precursor lesions and presents in the form of distinct growth patterns (Chap. ▶ 147). Biology of disease, prognosticators, staging, risk factors, and pathogenic pathways of gallbladder carcinoma are treated in more detail in Chaps. ▶ 148 and ▶ 149. Apart from the common ordinary adenocarcinoma of the gallbladder, several rare variants with other differentiation patterns are recognized, including mucinous, signet ring cell, and squamous cell carcinomas (Chaps.  $\triangleright$  150 and  $\triangleright$  151). The gallbladder is also the site of rare cystic and mixed neoplasms, such as cystadenoma and cystadenocarcinoma (Chap. ▶ 152). As outlined in Chap. ▶ 153, the gallbladder can give rise to a spectrum of adenomatous, borderline, and dysplastic lesions. Of differential diagnostic importance is the observation of various types of hyperplastic and metaplastic lesions in the gallbladder mucosa (Chap. ▶ 154). Similar to the bile duct system and liver, the gallbladder is a well-known origin of diverse types of neuroendocrine tumors, mesenchymal neoplasms, malignant melanoma, and a wide spectrum of other, however very rare, neoplasms (Chaps.  $\triangleright$  155,  $\triangleright$  156, and  $\triangleright$  157). A broad array of reactive, inflammatory, and noninflammatory alterations of the gallbladder can result in mass lesions that mimic neoplasms, and in particular gallbladder cancer (Chaps. ≥ 158, ≥ 159, ≥ 160, and ≥ 161). A rare group of malignant and benign tumors takes its origin in the cystic duct (Chap. ▶ 162).

Part 30 refers to a heterogeneous group of tumorous and tumor-like peritoneal lesions that may involve the liver surface. They include several primary carcinomas and other malignancies, pseudomyxoma peritonei, gliomatosis peritonei, and various forms of metaplasia, granulomas, endometriosis, and deciduosis (Chap. ▶ 163).

Part 31 is the first part of the textbook referring to aspects of general pathology of hepatobiliary tumors, specifically etiology and pathogenesis of hepatocellular carcinoma (HCC). A first chapter discusses in depth inflammatory and toxic causes, in particular the role of hepatitis virus infections, fatty liver and steatohepatitis, and nutritional and other toxins in hepatocarcinogenic pathways (Chap. ▶ 164). The following chapter (Chap. ▶ 165) discusses HCC that arises in the setting of inborn errors of metabolism, in particular various forms of chronic hepatic iron overload. Chapter ▶ 166 focuses on chromosomal alterations, oncogenes, tumor suppressors, and associated signaling networks that are involved in tumorigenesis, while Chap. ▶ 167 puts emphasis on the roles of transcription factors, regulators of growth and apoptosis, and telomere homeostasis. Finally, Chap. ▶ 168 is an overview on the etiologic and pathogenic significance of epigenetic mechanisms in hepatocarcinogenesis (the epigenome).

Part 32 contains an important chapter on the general pathology of structural and functional nuclear changes in hepatobiliary cancer. In Chap. ▶ 169,

Preface xv

relevant diagnostic and theoretical aspects of nuclear and nucleolar abnormalities, anaplasia, and chromatin alterations are addressed. Numerous types of structural abnormalities of cancer cell nuclei are directly associated with functional disorders of nuclear homeostasis, DNA replication, cell division, and organization of chromatin superstructures during interphase. Abnormal heterochromatin generation, a deranged production of euchromatin strings, malposition of interphase chromosomes, and anomalies of intranuclear chromosome movements are hallmarks of nuclear function failure in cancer cells (Chap.  $\triangleright$  170).

Part 33 addresses mitochondrial structure and function in normal and malignant neoplastic cells. Apart from their central role in energy production, mitochondria play a significant role within carcinogenic pathways involving abnormal stress responses and deregulation of cell death pathways. This role has led to the "mitochondrial malignancy theory." Mitochondria hold a central position in apoptosis induction, but they also modulate cell shape and engage in complex interactions with other organelles. Cancer cells exhibit various types of structural abnormalities of mitochondria and may show changes in mitochondrial number, mitochondrial fission, and elimination of this organelle. Part of these alterations are associated with losses and mutations of mitochondrial DNA (Chap. ▶ 171).

The contents of Part 34 pertain to tumor growth and its regulation. Uncontrolled, progressive growth is a key feature of cancers. Net mass increase of tumors not only depends on cell proliferation, but also on cell loss caused by various forms of apoptosis and necrosis and the contribution of nonneoplastic tissues and cells accompanying neoplasms, in particular stroma, blood vessels, and leukocytes. Cell proliferation in liver cell cancer reflects features of normal liver regeneration which is therefore discussed in some detail. The aberrant proliferation of liver cancer cells is related to deranged functions of factors orchestrating cell division, checkpoint regulators, proteins involved in DNA synthesis, proteins of the mitotic apparatus, and the numerous components of the cytoskeleton. Similar to the regenerating liver, growth of liver neoplasm is regulated by numerous growth factors and their receptors, including factors produced by platelets and antagonists of proliferation. Furthermore, hepatobiliary cancers reveal abnormal expression patterns of proteins that control entry into and passage through the cell division cycle, including cyclins and cyclin-dependent kinases. Finally, regulation of tumor growth strongly depends on various epigenetic mechanisms, specifically on complex expression patterns of microRNAs and other RNA classes (Chaps.  $\triangleright$  172,  $\triangleright$  173,  $\triangleright$  174, and  $\triangleright$  175).

Part 35 informs the reader about important aspects of the necrobiology of hepatobiliary cancer. An intricate process to control tumor cell mass is apoptosis, a complex form of tightly controlled cell death. Growth caused by proliferation is also counteracted by necrosis which, in contrast to traditional views, is a controlled process rather than a passive phenomenon. In liver cancer, apoptosis can be assessed by immunohistochemical and molecular methods. In addition to classical apoptosis, necrobiologic processes active in cancer also include various forms of cell death related to, but not identical with, apoptosis. These pathways may play a significant role for future novel

xvi Preface

therapies (Chaps. ▶ 176 and ▶ 177). A special chapter is dedicated to the pathophysiology of classical (passive) necrosis in comparison with regulated necrosis (necroptosis). The latter involves a complex signaling platform, the necrosome, a molecular machine that senses ATP depletion and transmits this signal into kinase execution switches (Chap. ▶ 178). An important role in cancer cell biology is played by autophagy, a process involved in the maintenance of cell and tissue homeostasis, control of the protein composition of cells, aging, senescence, and neoplastic transformation. Autophagy is the instrument to eliminate altered proteins, damaged or superfluous organelles, and pathogens, and is a complex system connected with inflammasome function, inflammation, immunogenic cell death, and cell senescence. Special forms of autophagy in cancer cells include mitophagy and nucleophagy (Chap. ▶ 179).

Part 36 covers several important aspects of cancer invasion and metastasis. Invasion and metastatic spread of cancer cells involve a highly complex sequence of events that comprise tumor cell individualization, tumor cell polarization, migration, and the acquisition of a secretory phenotype, with release of histolytic enzymes. It is not yet fully known how these features are acquired by cancer cells in a seemingly concerted fashion. For being able to locomote, cancer cells, similar to leukocytes, must be able to undergo shape change and polarization, a process that requires numerous cytoskeletal components and specific polarity proteins. The invasive process strongly depends on the generation of invadosomes, including podosomes and invadopodia, matrix-degrading adhesive, and actin-dependent dynamic cellular structures or "organelles" that can also extend through endothelial linings and mediate extravasation of tumor cells. An important role for the invasion of carcinomas is the distinct tumor stroma. Stroma is composed of cancer-associated fibroblasts/CAFs, myofibroblasts, mesenchymal stem cells, stellate cells, blood vessel cells, extracellular matrix, and several classes of infiltrating leukocytes. The interaction of stromal cells with cancer cells affects invasive functions and modulates epithelial-mesenchymal transition (Chaps. ▶ 180, ▶ 181, ▶ 182, and ▶ 183). Metastatic spread of cancer cells, preceded by invasion, is a process that depends on the construction of premetastatic niches, the expression of distinct prometastatic genes and metastasis suppressors, on numerous microRNAs, and on the exchange of cellular information through exosomes and other vehicles that transfer signal cargo and extracellular nucleic acids (Chap. ▶ 184).

Part 37 is reserved for a distinct tumor tissue that affects numerous biologic functions of neoplasms, i.e., tumor stroma. Stroma forms a specific microenvironment that critically regulates the development and behavior of malignant neoplasms. Stromal cells interact with tumor cells directly, in part through cell fusion, and via molecular signals, resulting in a complex signal platform that expands in parallel with tumor growth. Stroma regulates tumor growth, differentiation, invasion, and metastatic spread. The various types of leukocytes present in stroma, in particular tumor-associated macrophages, myeloid-derived suppressor cells, lymphocytes, and neutrophils, create a unique inflammatory microenvironment which, through chemokines and other signal substances, significantly affects tumor biology (Chap. ▶ 185).

Preface xvii

Part 38 shows how tumor angiogenesis functions and how it is an essential process in many aspects of liver tumor invasion and progression. Angiogenesis, the formation of new tumor blood vessels, is a critical mechanism for the development and progression of hepatobiliary tumors, which are often highly vascular lesions. In contrast to normal tissues, tumor blood vessels often form highly atypical branching patterns, with irregular diameters and abrupt changes from large to small diameters. The cells mediating tumor angiogenesis are endothelial cells and auxiliary cells that modulate the biology of endothelial cells, in particular perivascular cells, stromal cells, and tumor-associated macrophages and other leukocytes. As in normal tissue, initiation and progression of angiogenesis in tumors involve the action of numerous angiogenic factors, but neoplasms also produce several antiangiogenic factors. Tumor angiogenesis is modulated by epigenetic mechanisms, mainly microRNAs expressed by tumor cells and stromal cells (Chap. ▶ 186). This chapter is supplemented by a chapter that addresses basic questions of vasculogenesis, angiogenesis, and lymphangiogenesis (Chap.  $\triangleright$  187).

The last part of the present book, Part 39, provides a summary of current systems of tumor staging. As with other cancers, staging of hepatobiliary cancers is critical for prognostication and optimal treatment planning. Staging is a complex task that depends on multiple factors. In recent years, several staging systems have been developed and markedly improved the methods to arrive at optimal risk stratification procedures. Apart from hepatocellular carcinoma, highly reproducible staging systems have been developed for extrahepatic and intrahepatic cholangiocarcinoma and for hepatoblastoma (Chap. ▶ 188).

Arthur Zimmermann Bern, Switzerland

#### **Acknowledgments**

It is both a great honor and pleasure to thank all those who encouraged me to be involved in this endeavor and offered me continuous advice, support, and help. In fact, the present textbook could only be created thanks to numerous persons who have contributed in several ways to make this work possible. First of all, I express my deep gratitude to my former and venerated teacher in pathology, the late Professor Hans Cottier of the Berne University who, when I was still a beginner in surgical pathology and research, inspired me and filled his young disciple with enthusiasm to engage in a mode of thinking that was strongly based on science. Specifically, it was he who so clearly showed me the uppermost significance of basic science in pathology and, by doing this, kindled my strong desire to follow this path of thinking and acting. Later in my professional life as a pathologist I had the privilege to chair the Berne Institute of Pathology together with an outstanding partner, Professor Thomas Schaffner, whose critical mind in science and the highly stimulating interaction with him as a really good friend will remain unforgettable and irreplaceable. In regard to my early and continued interest in hepatology, I am indebted to all those who uncovered this fascinating field for me, in particular Professors Preisig, Reichen, Lauterburg, and Dufour of the Berne University. In the field of surgery, I have profited a lot from intense contacts with Professors Blumgart and Büchler, with whom I had the great chance to cooperate in various scientific projects and who converted me to bring hepatobiliary pathology to the bedside. In the field of pediatric hepatobiliary oncology, I would like to acknowledge the stimulating influence and help of all my professional friends of the international SIOPEL study group, with whom I had a very fruitful interaction for so many years. Having had access to large numbers of hepatobiliary tumor specimens deserves special thanks for the enthusiasm and tireless work of the laboratory crew of the Berne pathology institute, in particular I remain indebted to the wonderful support I have received from the histopathology staff with its chief, Mary Economou. A hearty thanks also goes to all who so often served me with special methods in diagnostic pathology, particularly Andreas Kappeler and his crew. A central place in the project of this book is held by my outstanding scientific secretary, Erna Kramel. I am highly grateful for the excellent secretarial help and the skillful illustration work performed by Erna, who spared no effort in seeing the preparatory work being completed. I also deeply acknowledge the outstanding librarian work of Sibylle Graf and Irene Marconi who, during many years, competently

xx Acknowledgments

provided me with large numbers of scientific articles and books and performed literature search with great motivation and innovation. I extend my appreciation to Alexander Grossniklaus for his continued help in matters of informatics. The significance of the contributions of all these persons cannot be overestimated. The author has been fortunate in having a continued and outstanding relationship with his publisher and his Springer partners associated with and enthusiastically accompanying him in this fascinating project, in particular Daniela Graf who offered continued excellent support and advice. Finally, the creation of this work would have been inconceivable without the continued interest, help, and compassion of my wife, Geneviève, who endured much in all these years of hard work and who accompanied me with great loyalty in this endeavor.

#### **Contents**

#### Volume 1

Part Precu	I Tumors of the Hepatocyte Lineage and Its ursors	1
1	Liver Cancer: Stem and Progenitor Cells	3
2	Hepatocellular Carcinoma (Ordinary Hepatocellular Carcinoma)	27
3	Immunohistochemistry of Hepatocellular Carcinoma	65
4	Invasion Patterns and Metastatic Patterns of Hepatocellular Carcinoma	91
5	Secondary Alterations of Hepatocellular Carcinoma	121
6	Hepatocellular Carcinoma: Prognostic Factors	151
7	Precursor Lesions of Hepatocellular Carcinoma	167
8	Early or Small Hepatocellular Carcinoma	195
9	Clear Cell Hepatocellular Carcinoma and Related Tumors	215
10	Steatotic and Steatohepatitic Hepatocellular Carcinomas and Related Neoplasms	229
11	Hepatocellular Carcinomas with Increased Stromal Reactions	251
12	Inflammatory Hepatic Carcinomas	261
13	Variants of Hepatocellular Carcinoma	275
14	Hepatocellular Carcinoma with Progenitor Cell Features	293
15	Ectopic Hepatocellular Carcinoma	305
16	Pediatric Hepatocellular Carcinoma	317

xxii Contents

17	Hepatoid Carcinomas (Adenocarcinomas with Hepatoid Features)	329
18	Fibrolamellar Hepatocellular Carcinoma	335
19	Hepatoblastoma and the Hepatoblastoma Family of Tumors	357
20	Variants of the Hepatoblastoma Tumor Family	395
21	Hepatoblastomas: Biology of Disease and Prognostic Factors	405
22	Hepatoblastoma Family of Tumors: Risk Factors and Pathogenic Pathways	411
23	Nested Stromal-Epithelial Tumor of the Liver	435
24	Hepatocellular Adenoma	443
25	Variants of Hepatocellular Adenoma	473
26	Combined Hepatocellular-Cholangiocarcinoma	481
Part	II Tumors of the Cholangiocyte Lineage	499
27	Hilar/Perihilar Cholangiocarcinoma (Klatskin Tumor)	501
28	Extrahepatic Cholangiocarcinoma: Carcinoma of the Middle and Distal Common Bile Duct (Middle and Lower Bile Duct Carcinomas)	527
29	Intrahepatic Cholangiocarcinomas (ICCs)	549
30	Biology of Disease and Prognostic Factors of Cholangiocarcinomas	587
31	Etiology and Pathogenic Pathways of Cholangiocarcinoma	605
32	Pediatric and Adolescent Cholangiocarcinoma and Related Lesions	639
33	Intraductal Neoplasms of the Biliary Tract	645
34	Cholangiocarcinomas Arising in Cystic Lesions of the Hepatobiliary Tract and in Pancreaticobiliary Maljunction	669
35	Bile Duct Carcinomas with Marked Extracellular or Intracellular Mucin Accumulation: Mucinous and Signet Ring Cell Carcinomas	687
36	Special Variants of Cholangiocarcinoma	699
37	Clear Cell Carcinomas of the Bile Ducts	
- ,		, . ,

Contents xxiii

38	Bile Duct Tumors with Oncocytic Features	721
39	Mucinous Cystic Neoplasms (MCN) and Related Cystic Neoplasms of the Hepatobiliary Tract	727
40	Benign Epithelial Tumors and Hamartomas of the Biliary Tract	749
41	Tumors and Tumor-like Lesions of Peribiliary Glands	779
Part Epitl	III Hepatobiliary Tumors Derived from Other helial Lineages	789
42	Carcinomas with a Squamous Cell Lineage	791
43	Hepatobiliary Tumors with Acinar Cell and Adenoid Cystic Cell Lineages	813
44	Rare Malignant and Semimalignant Epithelial Neoplasms of the Biliary Tract	819
45	Epithelial Tumors of the Liver of Uncertain Lineage	831
Нера	IV Mixed Epithelial-Mesenchymal Tumors of the atobiliary Tract (Tumors with Epithelial-Mesenchymal sition)	841
46	Hepatobiliary Carcinosarcomas and Related Neoplasms	843
47	Giant Cell Tumors of the Hepatobiliary Tract	859
48	Mixed Tumors of the Liver and Related Neoplasms	865
Vol	ume 2	
Part	V Vascular Tumors of the Hepatobiliary Tract	873
49	Cavernous and Small Vessel Hemangiomas of the Hepatobiliary Tract	875
50	Variants of Hepatobiliary Angiomatous Tumors	907
51	Hepatic Epithelioid Hemangioendothelioma	927
52	Infantile Hepatic Hemangioma/Hemangioendothelioma	937
53	Variants of Hemangioendotheliomas of the Hepatobiliary Tract	959
54		
	Angiosarcoma of the Liver	965
55	Angiosarcoma of the Liver  Pediatric Hepatic Angiosarcomas	965 989

xxiv Contents

57	Glomus Tumor and Glomangiopericytic Tumors of the Liver	2
<b>=</b> 0		
58	Tumor-Like Vascular Malformations	
59	Bacillary Angiomatosis and Peliosis Hepatis 1023	3
	VI Tumors and Tumor-Like Lesions of Lymph Vessels e Hepatobiliary Tract	9
60	Lymph Vessel Tumors of the Hepatobiliary Tract 104	1
	VII Solitary Fibrous Tumor and Tumors with a angiopericytoma-Like Pattern	5
61	Solitary Fibrous Tumor/Hemangiopericytoma of the Liver	7
Part	VIII Mesenchymal Tumors of the Hepatobiliary Tract 1069	9
62	Fibroblastoid and Myofibroblastoid Tumors of the Liver and Bile Ducts	1
63	<b>Leiomyomatous Tumors of the Hepatobiliary Tract</b> 108	7
64	Tumors of the Striated Muscle Cell Lineage: Hepatobiliary Rhabdomyosarcoma and Rhabdomyoma	7
65	Lipomatous Tumors, Liposarcoma, Benign Adipocyte-Containing Nonneoplastic Lesions, and Focal Fatty Changes of the Liver	3
66	Tumors of Osseous and Chondroid Cell Lineages 115:	5
67	Gastrointestinal Stromal Tumors (GISTs) of the Liver 116	1
68	Tumors and Tumor-Like Lesions of the Schwann Cell Lineage and the Nerve Sheath	3
69	Tumors of a Granular Cell Lineage	3
70	Synovial-Like Neoplasms (Synovial Sarcoma) of the Liver	5
71	Malignant Fibrous Histiocytoma of the Liver: A Neoplasm of the Undifferentiated High-Grade Pleomorphic Sarcoma Group	5
Part	IX Tumors of the Mesothelial Lineage	5
72	Hepatic Mesotheliomas and Related Neoplasms	7

Contents xxv

	Tumors of the Perivascular Epithelioid Cell (PEC) n	1239
73 1	Perivascular Epithelioid Cell Tumors	1241
Part X	I Hepatobiliary Tumors of Neuroendocrine	
Lineag	ges	1265
74 ]	Paraganglioma of the Hepatobiliary Tract	1267
<b>75</b> I	Neuroendocrine Tumors of the Hepatobiliary Tract	1279
	II Hepatobiliary Tumors with a Small Cell erentiated Lineage	1307
	Undifferentiated Small Cell Tumors of the Hepatobiliary Tract	1309
Part X	III Hepatobiliary Melanotic Tumors	1329
	Primary and Secondary Malignant Melanoma, and Other Melanotic Tumors, of the Hepatobiliary Tract	1331
	IV Hepatobiliary Tumors with a Rhabdoid neage	1357
	Malignant Rhabdoid Tumors and Tumors with Rhabdoid Features	1359
Part X	V Hepatobiliary Germ Cell Tumors	1385
79	Germ Cell Tumors of the Liver	1387
Part X	VI Hepatobiliary Myeloid Neoplasms	1411
<b>80</b> I	Myeloid Neoplasms with an Erythroid Cell Lineage	1413
<b>81</b> I	Myeloproliferative Syndromes and Thrombocythemia	1427
<b>82</b> I	Myeloid Neoplasms with Eosinophil Lineage	1455
83 5	Systemic Mastocytosis and Mast Cell Tumors	1469
	Acute Leukemias of Granulocytic, Erythroid, and Megakaryocytic Lineages	1487
<b>85</b> I	Myeloid Neoplasms with a Monocytoid Lineage	1511
	Malignant and Benign Extramedullary Tumor-Forming Myeloid Proliferations	1523
	Blastic Plasmacytoid Dendritic Cell Neoplasm	

xxvi Contents

Part	XVII Hepatobiliary Hodgkin's Disease
88	Hodgkin Lymphoma of the Hepatobiliary Tract 1541
Othe	XVIII Hepatobiliary Non-Hodgkin's Lymphomas, er Lymphoproliferative Disorders, and Neoplasms/ferations of the Dendritic and Histiocytic Systems 1577
89	B-Cell Non-Hodgkin's Lymphomas with a Small Cell to Intermediate Cell Phenotype
90	B-Cell Non-Hodgkin's Lymphomas with a Small Cell to Intermediate Cell Phenotype: Special Phenotypes 1601
91	B-Cell Non-Hodgkin's Lymphomas with a Blastic/Large Cell Phenotype
92	B-Cell Non-Hodgkin's Lymphomas with Lymphoplasmacytoid and Plasmacytic Features
93	Amyloidogenic Immunoglobulin Chain and Non-amyloidogenic Immunoglobulin Chain Diseases 1659
94	B-Cell Non-Hodgkin's Lymphomas Associated with Viral Infections
95	T-Cell Non-Hodgkin's Lymphomas
96	T-Cell Non-Hodgkin's Lymphomas: Variant Forms 1711
97	<b>Posttransplant Lymphoproliferative Disorders (PTLDs)</b> 1721
98	<b>Hepatobiliary Castleman Disease</b>
99	Lymphoid Hyperplasia and Pseudolymphomas of the Hepatobiliary Tract
100	Tumors of the Langerhans Cell System
101	Neoplasms of Histiocyte/Macrophage Lineage: Histiocytic Sarcoma and Similar Neoplasms
102	Tumors of the Dendritic Cell System
103	Histiocytic Syndromes
Volu	ume 3
	XIX Mesenchymal Hamartoma and Related blasms
104	Mesenchymal Hamartoma of the Liver

Contents xxvii

105	Undifferentiated (Embryonal) Sarcoma (UES)	1857
Part 2	XX Metastatic Liver Disease	1877
106	Metastatic Liver Disease: Pathological Aspects	1879
107	Metastatic Liver Disease: Colorectal Carcinomas	1907
108	Metastatic Liver Disease: Tumors Other Than Colorectal Carcinoma	1927
109	Metastatic Liver Disease: Secondary Alterations of Hepatic Metastases	1947
110	Metastatic Liver Disease: Hepatic Lymph Node Involvement	1965
111	Metastatic Liver Disease: Associated Liver Lesions	1973
112	Pathogenic Features of Liver Metastasis: Growth, Regrowth, Stem Cells, and Circulating Tumor Cells	1989
113	Pathogenic Features of Liver Metastasis: Mechanisms Involving Platelets, Tumor Stroma, Epithelial-Mesenchymal Transition, and the Premetastatic Niche	1997
114	Pathogenic Features of Liver Metastasis: The Complex Cancer Cell–Cancer Cell, and Cancer Cell–Microenvironment Interactomes	2019
115	Pathogenic Features of Colorectal Liver Metastasis: Prometastatic and Antimetastatic Genes	2035
	XXI Tumors and Tumor-Like Lesions of the Hepatic ments	2043
116	Tumors and Tumor-Like Lesions of the Hepatic Ligaments	2045
Part 2	XXII Nodular Hyperplastic Lesions of the Liver	2055
117	Focal Nodular Hyperplasia (FNH) of the Liver	2057
118	Nodular Regenerative Hyperplasia and Other Noncirrhotic Nodular Hyperplastic Lesions of the Liver	2091
Part 2	XXIII Hepatobiliary Pseudotumors	2117
119	Pseudotumors and Related Lesions of the Hepatobiliary Tract	2119

xxviii Contents

	XXIV Nonneoplastic Mass Lesions of the atobiliary Tract
120	Ectopias and Heterotopies as Tumor-Like Lesions of the Hepatobiliary Tract
121	Tumor-Like Lesions of the Hepatobiliary Tract: Anatomical Variations and Malformations Resulting in Hepatic Mass Effects or Focal Defects
122	Solitary Necrotic Nodule of the Liver
123	Hepatic and Perihepatic Involvement in Pneumokonioses and Other Mineral-Induced Diseases
124	Tumor-Like Lesions of the Hepatobiliary Tract Caused by Gallstones, Foreign Bodies, and Bile
125	Hepatobiliary Pseudotumors Consisting of Nonneoplastic Hematopoietic Cells and Cells of the Macrophage Lineage
126	Liver Abscesses as Pseudotumoral Lesions
127	Tumor-Like Lesions of the Hepatobiliary Tract Caused by Mycobacterial Infections: Tuberculosis and Lepra 2279
128	Tumor-Like Lesions of the Hepatobiliary Tract Caused by Actinomycosis, Nocardiosis, and Botryomycosis 2311
129	Tumor-Like Lesions of the Hepatobiliary Tract: Granulomatous Masses in Syphilis, Brucellosis, and Cat-Scratch Disease
130	Tumor-Like Lesions of the Hepatobiliary Tract: Specific Abscess-Forming Bacterial Infections
131	Tumor-Like Lesions of the Hepatobiliary Tract Caused by Fungal Infections
132	Tumor-Like Protozoal Infections of the Hepatobiliary Tract
133	Tumor-Like Parasitic Lesions of the Hepatobiliary Tract: Liver Flukes and Other Trematodes
134	Tumor-Like Parasitic Lesions of the Hepatobiliary Tract: Echinococcosis and Cysticercosis
135	Tumor-Like Parasitic Lesions of the Hepatobiliary Tract: Round Worm Infestations
136	Tumor-Like Parasitic Lesions of the Hepatobiliary Tract:  Pentastomiasis

Contents xxix

137	Tumor-Like Necroses of the Liver: Liver Infarct and Hepatic Pseudo-infarct (Zahn's Infarct)	2445
Part	XXV Reactive Cystic Lesions of the Liver	2455
138	Cystic Hepatobiliary Lesions Mimicking Cystic Neoplasms	2457
	XXVI Hepatobiliary Mass Lesions Caused by infectious Granulomas	2473
139	Tumor-Like Granulomatous Disorders of the Hepatobiliary Tract	2475
140	Hepatobiliary Manifestations of Chronic Granulomatous Diseases of Childhood	2501
	XXVII Hepatobiliary Mass Lesions Caused by osclerotic Lesions	2509
141	Tumor-Like Fibrosclerotic Lesions of the Hepatobiliary Tract	2511
	XXVIII Reactive Bile Duct Alterations Mimicking ry Tumors	2531
142	Reactive Bile Duct Alterations Mimicking Biliary Cancer: Inflammatory Conditions	2533
143	Reactive Bile Duct Alterations Mimicking Biliary Cancer: Eosinophilic Cholangitis/Cholangiopathy and Other Eosinophilic Disorders	2551
144	Mechanical and Anatomical Causes of Bile Duct Obstruction	2561
145	Bile Duct Compression and Stenosis Due to Anomalies and Acquired Disorders of the Splanchnic Arterial Tree and the Portal Venous System	2585
146	Postcholecystectomy Changes, Biliary Tract Mineralizations, and Intrabiliary Foreign Bodies	2609
Volu	ume 4	
	XXIX Tumors and Pseudotumors of the pladder	2623
147	Adenocarcinoma of the Gallbladder (Classical Gallbladder Cancer)	2625

xxx Contents

148	Adenocarcinoma of the Gallbladder: Biology of Disease, Prognosticators, and Staging
149	Adenocarcinoma of the Gallbladder: Risk Factors and Pathogenic Pathways
150	Variant Adenocarcinomas of the Gallbladder
151	Gallbladder Carcinomas Not Related to Glandular Epithelia
152	Cystic Epithelial Tumors and Mixed Neoplasms of the Gallbladder
153	Benign/Borderline Epithelial Tumors and Dysplastic Alterations of the Gallbladder
154	Hyperplastic Lesions and Metaplastic Changes of the Gallbladder
155	Neuroendocrine Tumors of the Gallbladder
156	Mesenchymal Tumors of the Gallbladder
157	Melanocytic, Neuroectodermal, Germ Cell, Rhabdoid, Perivascular Epithelioid Cell, Hemolymphatic, and Metastatic Tumors of the Gallbladder
158	Tumor-Like Inflammatory Changes of the Gallbladder 2827
159	Noninflammatory Tumor-Like Changes of the Gallbladder
160	Cysts of the Gallbladder
161	Tumor-Like Infections and Infestations of the Gallbladder (Infectious and Parasitic Pseudotumors)
162	Tumors and Tumor-Like Lesions of the Cystic Duct 2887
Part	XXX Peritoneal Tumors and Tumor-Like Lesions 2903
163	Tumors, Tumor-Like Lesions, and Metaplastic Lesions of the Peritoneum
	XXXI General Pathology of Hepatobiliary Tumors: ogy and Pathogenesis of Hepatocellular Carcinoma 2929
164	Etiology and Pathogenesis of Hepatocellular Carcinoma: Inflammatory and Toxic Causes
165	Etiology and Pathogenesis of Hepatocellular Carcinoma: Hepatocellular Carcinoma Associated with Inborn Errors of Metabolism and Hepatobiliary Malformations

Contents xxxi

166	Etiology and Pathogenesis of Hepatocellular Carcinoma: Chromosomal Alterations, Oncogenes, Tumor Suppressors, and Associated Signaling Networks	2987
167	Etiology and Pathogenesis of Hepatocellular Carcinoma: Transcription Factors, Signal Pathways Regulating Proliferation and Apoptosis, and Telomeres/Telomerases	3007
168	Etiology and Pathogenesis of Hepatocellular Carcinoma: Epigenetic Mechanisms	3029
	XXXII General Pathology of Hepatobiliary Tumors: ctural and Functional Changes of Nuclei	3041
169	Nucleus, Nuclear Structure, and Nuclear Functional Changes in Liver Cancer	3043
170	Nucleus, Nuclear Structure, and Nuclear Functions: Pathogenesis of Nuclear Abnormalities in Cancer	3071
	XXXIII General Pathology of Hepatobiliary Tumors: chondrial Biology	3089
171	Mitochondrial Biology in Hepatobiliary Tumors: Changes of the Cellular Energy Factory	3091
	XXXIV General Pathology of Hepatobiliary Tumors: vth and Its Regulation	3125
172	General Aspects of Liver Regeneration and Hepatobiliary Cancer Growth	3127
173	Growth Regulation in Hepatobiliary Cancer: Involvement of Growth Factors	3159
174	Growth Regulation in Hepatobiliary Cancer: Regulators of the Cell Division Cycle	3173
175	Growth Regulation in Hepatobiliary Cancer: Epigenetic Mechanisms	3203
	XXXV General Pathology of Hepatobiliary Tumors:	
	obiology of Liver Cancer	3215
176	Necrobiology of Liver Cancer: Apoptosis and Related Forms of Cell Death	
	Necrobiology of Liver Cancer: Apoptosis and Related	3217

xxxii Contents

179	Necrobiology of Liver Cancer: Autophagy and Cellular Senescence
	XXXVI General Pathology of Hepatobiliary Tumors: sion and Metastasis
180	Mechanisms of Invasion and Metastasis: General Aspects and the Role of Cell Junctions, Adhesion, and Extracellular Matrix
181	Mechanisms of Invasion and Metastasis: Cell Migration and Chemotaxis
182	Mechanisms of Invasion and Metastasis: Tissue Invasion
183	Mechanisms of Invasion and Metastasis: Role of the Stromal Liver Cancer Microenvironment, Epithelial-Mesenchymal Transition, and the Tumor Vascular Bed
184	Mechanisms of Invasion and Metastasis: Prometastatic and Antimetastatic Factors
	XXXVII General Pathology of Hepatobiliary Tumors: or Stroma
185	Tumor Stroma, Desmoplasia, and Stromagenesis 3409
	XXXVIII General Pathology of Hepatobiliary Tumors: ogenesis
186	Angiogenesis in Liver Cancer
187	Angiogenesis in Liver Cancer: General Aspects and Cellular Sources of Normal Angiogenesis
Part	XXXIX Staging of Liver Cancer
188	Staging of Hepatocellular Carcinoma, Hepatoblastoma, and Cholangiocarcinomas
Inde	x 3517

#### **About the Author**

Professor Arthur Zimmermann is an internationally known specialist in hepatobiliary tumor pathology. Following his training as MD and pathologist at the University of Berne, Switzerland, he worked in basic research for several years, focusing on tumor cell growth regulation, tumor cell locomotion, and cell cycle mutants of cancer cells. In surgical pathology, he analyzed more than 20,000 liver specimens, described new tumor entities, and was an author or coauthor of more than 500 publications in his field of interest. As chapter author, he participated in several well-known books on liver and biliary tract disease, including the 2010 edition of the WHO Classification of Tumors of the Digestive System, and was one of the editors of the book, Pediatric Liver Tumors (Pediatric Oncology Series, Springer). Professor Zimmermann developed the pathology review center for the multinational SIOPEL pediatric liver cancer treatment studies and was involved in the formulation of the new classification of pediatric liver tumors.

Part I

**Tumors of the Hepatocyte Lineage and Its Precursors** 

Contents		Regulation of Stem Cells by microRNAs and	1.5
Introduction	4	Other Noncoding RNAs	15
Stem Cells and Progenitor Cells in the Normal Liver: Pacemakers of Hepatic Ontogeny and Regeneration in the Injured Liver	5	by Microvesicles and Exosomes: Can Cancer Stem Cells "Infect" Other Cells and Transmit	15
Extrahepatic Stem Cells Home to Liver and Liver Cancer and Regulate the Respective	_	Systemic and Hepatic Mesenchymal Stem Cells and Stem-Like Cells: Modulation of HCC Biology	
Cell Biologies	5	Dormant Cancer Stem Cells	16
Identification of Liver Tumor Stem Cells: Morphology, Stem Cell Markers, and Their		References	16
Characteristics  Morphology  Markers of Stem/Progenitor Cells  CD133: Prominin 1 is a Crucial Molecule in  HCC-CSC Biology  Other Stem Cell Markers	6 6 6 7 7		
	/		
HCC-Associated Stem Cells: Self-Renewing Cells with Various Biological Functions Stem Cells in HCC: Significance for Invasion,	8		
Spread, Stroma Formation, and Angiogenesis	10 11		
CSCs in Tumor Recurrence and Metastatic Spread: RASCs, NASCs, and MASCs	12		
Tumor Stem Cells in Nontumorous Liver Tissue	13		
Cells with Stem Cell-Like Features in Special Variants of Hepatocellular Carcinoma and in	12		
Precursor Lesions	13		
Stem Cells in Hepatoblastomas	13		
<b>Biogenesis of HCC-Associated Stem Cells</b>	13		
and Liver Neoplasms Pathways from Primed Stem Cells to Differentiated	13		
Cells	14		

<sup>©</sup> Springer International Publishing Switzerland 2017 A. Zimmermann, *Tumors and Tumor-Like Lesions of the Hepatobiliary Tract*, DOI 10.1007/978-3-319-26956-6\_1

#### **Abstract**

Stem cells are involved in hepatocarcinogenic pathways. In the normal liver, hepatic stem cells are self-renewing cells dwelling in distinct compartments termed stem cell niches. These niches are distinct microenvironments which maintain the balance of slow selfrenewal of quiescent stem cells and their priming to become hepatocytes and other differentiated cells. Hepatic stem cells possess plasticity, allowing them to undergo lineage diversification. In hepatocarcinogenesis, tumor stem cells mimick certain features of their normal counterparts. Tumor stem cells in hepatocellular carcinoma (HCC) can act as tumor-initiating cells and tumor-propagating cells that express hepatic stem cell markers and show distinct gene expression profiles. Apart from their role in carcinogenic pathways, cancer stem cells in HCC are considered to play an important role in tumor recurrence and metastasis. The expression of stemnessrelated markers in HCC confers an aggressive phenotype in these neoplasms. Tumor stem cells in HCC modulate several processes, including maintenance of slowly cycling clonogenic cells having longevity, induction of distinct growth features, and tumor progression. Following removal of HCC and other liver cancers, tumor stem cells can remain in the liver and enter the bloodstream to settle in new stem cell niches, including metastatic niches.

#### Introduction

Stem cells participate in tumorigenic pathways of numerous neoplasms and have therefore anticipated to be involved in hepatocarcinogenesis. Tumor stem cells (cancer stem cells, CSCs) in hepatocellular carcinoma (HCC) drive carcinogenic pathways and maintain a growing cell population in addition to proliferating nonstem cancer cells, but stem cells are also considered to play a

significant role in HCC metastasis and recurrence. Furthermore, and similar to normal stem cells, CSCs exhibit the features of plasticity in regard to their lineage fating, an element that plays an important role in tumor heterogeneity and progressive phenotypic change, e.g., in metastases Meacham and Morrison (reviews: 2013; Sukowati and Tiribelli 2013). Recent research on liver cancer stem cells has also been undertaken with the aim of shedding light on novel treatment directions (reviews: Lee et al. 2009; Pang and Poon 2012). Generally, it is proposed that the expression of stemness-related markers in HCC confers an aggressive phenotype to these neoplasms. As in other organs and tissues, hepatic stem cells are self-renewing cells that dwell in distinct compartments termed stem cell niches, specific microenvironments which provide conditions for maintaining a balance of slow selfrenewal of the mostly quiescent stem cells and their priming to become differentiated cell lineages. It is assumed that, following removal of HCC and other malignancies, stem cells involved in maintenance of a neoplastic phenotype remain in the liver and can enter the bloodstream to settle in new stem cell niches, including metastatic niches.

Selected References Sell (1993), Alison and Lovell (2005), Shupe and Petersen (2005), Wu and Chen (2006), Kitisin et al. (2007), Dan and Yeoh (2008), Sell and Leffert (2008), Zou (2008), Alison et al. (2009), Chiba et al. (2009), Lee et al. (2009), Mishra et al. (2009), Yao and Mishra (2009), Darwiche and Petersen (2010), Marquardt and Thorgeirsson (2010), Oliva et al. (2010), Teufel and Galle (2010), Lau et al. (2011), Liu et al., (2011a), Tong et al. (2011), Wang and Jacob (2011), Alisi and Baffet (2012), Ji and Wang (2012), Rountree et al. (2012), Shen and Cao (2012), Yoon (2012), Blechacz and Mishra (2013), Dong et al. (2013), Jia et al. (2013), Oishi et al. (2014), Sainz and Heeschen (2013), Yamashita and Wang (2013), Yamashita and Kaneko (2014).

# Stem Cells and Progenitor Cells in the Normal Liver: Pacemakers of Hepatic Ontogeny and Regeneration in the Injured Liver

Stem cells are critically involved in liver ontogenesis (reviews: Kamiya et al. 2006; Navarro-Alvarez et al. 2010). After fating of primordial endodermal cells to become liver, hepatic specification produces bipotential hepatic stem cells or hepatoblasts which can differentiate into either hepatocytes or cholangiocytes (reviews: Sangan and Tosh 2010; Parveen et al. 2011; Shin and Monga 2013; Kamiya and Inagaki 2015). Hepatic specification is a highly complex process that not only depends on fated endodermal cells but also on an interaction of several cell types located to endoderm, including the endothelial cell niche which promotes hepatic endoderm expansion and is directly required for liver speciation (Han et al. 2011b). In regard to regeneration of hepatocytes, the liver is in a special situation insofar as most of the cells that repopulate undamaged parenchyma after cell loss (e.g., resection) are the mature hepatocytes themselves, and not stem/progenitor cells, as hepatocytes are in a G0 phase of cycle and can be rapidly recruited into a cell division cycle. Stem cells play a significant role in regeneration of the liver in case of hepatocyte damage, i.e., when metabolic injury and decreased viability precludes a regenerative response of hepatocytes. Two types of hepatic stem cells have been defined, viz. fetal hepatic stem cells and stem cells of the adult liver (oval cells; progenitor cells). Fetal hepatic stem cells can maintain their self-renewal capability in the developing liver (Zheng and Taniguchi 2003). In adult livers, hepatic progenitor cells are present in small numbers and are quiescent stem cells with a low proliferative rate, representing a reserve or "emergency" compartment that is activated in case of marked hepatocyte damage or massive hepatocyte loss (Sharma et al. 2006; Gaudio et al. 2009; "ghosts in the machine"; Darwiche and Petersen 2010). In the normal adult, liver stem cells mainly reside in peripheral parts of portal tracts, in the

compartment of canals of von Hering, and ductules (Vessey and de la Hall 2001), but a subset of liver progenitor cells originates from HNF1beta(+) biliary duct cells (Rodrigo-Torres et al. 2014). The progenitor cell response comes in four components, viz. activation/priming, proliferation, migration, and differentiation. In the course of stem cell-mediated regeneration, primed stem cells and their progeny, oval cells in rodents and progenitor cells in humans, form ductular profiles (ductular reaction) that deeply invade the damaged parenchyma and differentiate into focally arranged cell clusters that will generate hepatocytes (reviews: Fausto and Campbell 2003; Vestentoft 2013; Katoonizadeh et al. 2014). These clusters can also form narrow intraparenchymal ductular structures which have been termed parenchymal ductules. The cells forming these ductules have the same immunophenotype as resting stem cells (Papp et al. 2014). Stem cells isolated from fetal liver have the capacity to repopulate up to 10 % of normal liver, while progenitor cell lines from embryonal and adult liver have no significant repopulation activity (review: Dabeva and Shafritz 2003).

## Extrahepatic Stem Cells Home to Liver and Liver Cancer and Regulate the Respective Cell Biologies

In addition to resident hepatic stem and progenitor cells, stem cells acting in the liver can also be recruited from extrahepatic organs and tissues. Induced pluripotent stem cells (iPSCs) develop through reprogramming of somatic cells to a pluripotent state by a complex induction process. iPSCs resemble embryonic stem cells/ESCs in several aspects (review: Chiang et al. 2013). iPSCs can, e.g., originate from mesenchymal cells (including mesenchymal stem cells), fibroblasts, adipocytes, and several types of bone marrow derived cells. As iPSCs can undergo tridermal priming/fating and differentiation, they can give rise to several types of hepatic cells. Embryonal

fibroblasts can be reprogrammed by hepatocyte nuclear factor 1 alpha and Foxa3 to bipotential hepatic stem cells (Yu et al. 2013b). Circulating bone marrow and/or adipose tissue-derived hematopoietic and mesenchymal stem cells (MSC) are capable to give rise to hepatic cells enabled to act in hepatic regeneration (Petersen et al. 1999; Grompe 2005; Oh et al. 2007; Guest et al. 2010; Ishikawa et al. 2010; Gong et al. (2013a, b); Stock et al. 2014). MSCs form a heterogeneous population of fibroblastoid cells surrounding blood vessels and are concentrated in the bone marrow and in adipose tissue, from where they can emigrate, circulate in the blood, and home to diverse tissues. Both hematopoietic stem cells and MSCs can home to the liver after hepatocyte loss, populate the remaining liver, and give rise to cells that can regenerate liver. MSCs can also home to tumors, including HCCs, where they exert diverse functions, including contribution to tumor cell development and expansion, generation of metastatic niches, neovascularization, and immunosuppression (review: Reagan and Kaplan 2011). Human fetal hepatic stem/progenitor cells are distinct from, but closely related to, hematopoietic stem cells (Chen et al. 2013d). Bone marrow-derived MSCs homing to liver tumors can, through interaction with tumor cells, increase tumor cell apoptosis and inhibit metastatic spread (Hang and Xia 2014). Homing of hematopoietic stem cells to the liver after hepatectomy requires hepatocyte growth factor (HGF) and stroma-derived factor-1 (SDF-1) (Lehwald et al. 2014), and hepatic differentiation of human adipose tissue-derived stem cells is promoted by distinct microRNAs (Davoodian et al. 2014).

#### Identification of Liver Tumor Stem Cells: Morphology, Stem Cell Markers, and Their Characteristics

#### Morphology

In the normal and regenerating liver, at least part of the stem or progenitor cells can morphologically be identified (Alison et al. 2001; Forbes et al. 2002; Fausto and Campbell 2003; Strain et al. 2003; Zheng and Taniguchi 2003;

Kuhlmann and Peschke 2006; Bird et al. 2008). Stem cells, which in part are located in the lining of bile ductules, from where they can be recruited and mobilized, have morphological features of oval cells, small basophil cells, and transition cells between cholangiocytes and hepatocytes. Recently, it was reported that also ballooned cells forming Mallory-Denk bodies may have features of progenitor cells (French et al. 2013). Numerous investigations provided strong evidence that hepatic cancer stem cells (CSC) can be isolated from HCC (Chiba et al. 2006; Ma et al. 2007; Kamohara et al. 2008; Sell 2008; Sell and Leffert 2008; Lingala et al. 2010; Tomuleasa et al. 2010; review: Ji and Wang 2012).

#### **Markers of Stem/Progenitor Cells**

CSC are characterized by several markers, including CD13, CD24, CD44, CD90, CD133, EpCam (CD326), OV6 (Terris et al. 2010), and numerous others (Table 1).

Stem cells or stem-like cells can also be isolated from HCC cell lines in vitro via isolation of the side population (SP) cells, a cell population

**Table 1** Markers for hepatocellular carcinoma-related cancer stem cells

capable to exclude Hoechst 33343 dye owing to the capacity of these cells with sufficient ATP-binding cassettes to transport the Hoechst dye out of the cell, while non-SP cells cannot. Due to this difference, tumor SP cells can then be separated from the other cell populations by use of cell sorting, and the resulting SP cells tested for typical stem cell features. Results of such analyses showed that SP cells contain a CSC cell population (review: Sell and Leffert 2008). However, not all cells of side population cell fractions have the characteristic features of CSCs (Nakayama et al. 2014). HCC-CSCs can also be identified by their molecular makeup. Stem cell features of HCC cells are associated with wellestablished gene alterations conferring an abnormal growth and apoptosis behavior in HCC, such as TP53 mutations (Woo et al. 2011).

## CD133: Prominin 1 is a Crucial Molecule in HCC-CSC Biology

CD133 (prominin 1; PROM1) is a well-known stem cell marker (Mizrak et al. 2008; Wu and Wu 2009; Li 2013) which is involved in cell differentiation and polarity (Fargeas et al. 2011), organ morphogenesis (Anderson et al. 2011), cell surface dynamics (Mak et al. 2011), iron homeostasis of various cells (Bourseau-Guilmain et al. 2011), and interactions between cancer cells and stromal cells (Moriyama et al. 2010; Akita et al. 2013a, b). CD133 is expressed in subsets of liver stem cells (Rountree et al. 2011). These cells display a greater colony-forming efficiency and a higher proliferative output (Ma et al. 2007). Expression of CD133 also characterized part of HCC stem cells (Suetsugu et al. 2006; Ma et al. 2008; Song et al. 2008; Yoshikawa et al. 2009; Suetsugu et al. 2010; Zhang et al. 2011; Tsai et al. 2012; Ma 2013), and CD133+/CD44+ cells in HCC are considered to be true cancer stem/progenitor cells that are involved tumor maintenance in and chemoradioresistance (Zhu et al. 2010; Piao et al. 2012). There is evidence that the presence of CD133+ stem cells in HCCs is related to the etiology of HCC. In an area endemic for HBV

virus (Taiwan), CD133 expression in HCC was negatively associated with the presence of HBsAg, implicating a non-HBV viral origin of CD133+ HCC (Yeh et al. 2009). In HCC cell lines, expression of CD133 markedly affects the biological features of the tumor cells. Silencing of CD133 expression impaired in vitro HCC cell proliferation, formation of tumor spheres, colony formation, and in vivo tumorigenicity in immunodeficient mice. In addition, knockdown of CD133 reduced cells in G0/G1 and increased tumor cell apoptosis through modulation of Bcl-2 and Bax (Lan et al. 2013). CD133 expression predicted poor disease-free survival independently of p53 expression (Yeh et al. 2009). Expression of CD133 in HCC cells is a predictor of the effectiveness of S1+ pegylated interferon alpha-2b therapy (Hagiwara et al. 2011). CD133 is also expressed in cholangiocarcinoma and gallbladder carcinoma cells (Shi et al. 2010; Fan et al. 2011; Iwahashi et al. 2013), and this expression is correlated with positive tumor margin status and lymph node metastasis (Leelawat et al. 2011).

#### Other Stem Cell Markers

A well-known second hepatic stem cell marker, OV6, is also present in HCC-associated CSC. OV6+ HCC stem cells are tumor-initiating cells that possess a high capacity to form tumor spheroids in vitro and give rise to tumors in severe combined immunodeficient (SCID) mice (Yang et al. 2012). CD90 characterizes a stem cell lineage in liver cancer, and cells expressing CD90 express several hundred other genes, including glypican-3 (Ho et al. 2012). A further protein expressed in part of stem and progenitor cells is c-Kit (CD117). In one investigation, c-Kit expression was detected in only 2.3 % of HCC, suggesting that c-Kit is not significantly overexpressed in HCC cells (Becker et al. 2007). In another study, immunoreactivity for c-Kit was detectable in 70 % of HCC with different degrees of intensity, illustrating highly variable results in regard to the expression of this stem cell marker in HCC. However, c-Kit reactivity was also found in 90 % of peritumoral cirrhotic and noncirrhotic liver tissue. Similarly, c-Kit

mRNA was identified in 83 % of HCC and 90-100 % of surrounding tissue (Mansuroglu et al. 2009). Sal-like protein 4/SALL4, a member of a family of zinc finger transcription factors, modifier of somatic cells and regulator of organogenesis and pluripotency, is a stem cell marker. SALL4 is expressed in normal human hepatic stem cells, hepatoblasts, HCC, cholangiocarcinoma, and mixed cholangiocarcinoma but is silenced in the normal adult liver. Its expression correlates with cell and tumor growth and an aggressive HCC phenotype (Yong et al. 2013), whereas its suppression results in slowed tumor growth and tumor cell differentiation (Oikawa et al. 2013). SALL4 is a marker for a progenitor subclass of HCC with an aggressive phenotype (Yong et al. 2013). Epithelial cell adhesion molecule (EpCAM) is expressed in several cancer and HCC stem cells and in tumor-initiating cells (TIC) (Yamashita et al. 2009; Terris et al. 2010; Imrich et al. 2012). In subsets of HCC stem cells, EpCAM is coexpressed with another stem cell marker, CD133, and cells with this phenotype represent tumor-initiating cells (TIC; Chen et al. 2012). In a recent investigation, it was shown that the two CSC markers of HCC, EpCAM and CD90, are independently expressed, in that EpCAM positivity characterized epithelial CSC, while CD90+ CSC had features of vascular endothelial cells, suggesting that more than one type of hepatic CSC may occur together and fulfill different functions (Yamashita et al. 2013). Liver cancer stem cells are also characterized by positivity for the Thomsen-Friedenreich core-1 protein, CD176 (Lin et al. 2011). Stemness features of HCC cells were analyzed by analysis of two stem cell characteristics, low proteasome activity and low intracellular reactive oxygen species/ROS. Isolated HCC with these two features demonstrated asymmetric divisions, tumorigenicity and a metastatic phenotype in immunodeficient mice, upregulated chemokine-related genes, and facilitation of macrophage migration in vitro (Muramatsu et al. 2013), suggesting that HCC stem cells not only affect tumorigenesis as such but also features of spread and interactions with tumor-associated cells such as macrophages. In HCC tumor cell lines kept in culture, a minor cell population expresses intercellular adhesion molecule 1 (ICAM-1). These ICAM-1-positive cells, which were also detected in human HCC and as circulating tumor cells in HCC patients, have greater sphere-forming and tumorigenic properties and represent tumor stem cells (Liu et al. 2013a). A second molecule involved in the regulation of cell adhesion is dysadherin. Dysadherin expression in liver cells characterizes a stem-like cell phenotype (Park et al. 2011). Part of HCC progenitor/stem cells are reactive for the 140 kDa isoform of neural cell adhesion molecule (NCAM) (Tsuchiya et al. 2011). A novel HCC stem cell marker is Delta-like 1 protein (Dlk-1), a surface protein expressed on fetal hepatic stem/progenitor cells but absent from mature hepatocytes in neonatal and adult rodent livers (Yanai et al. 2010). Overall, Dlk-1 is expressed in 20 % of all HCC but is even more frequently detectable in HCC diagnosed in young patients (Nishina 2012). Granulin-epithelin precursor (GEP) is a hepatic oncofetal protein expressed in fetal livers, but not in normal adult livers. Fetal liver cells containing GEP coexpress the stem cell-related signaling molecules betacatenin, Oct4, Nanog, Sox2, and Dlk-1 and the CSC markers CD133, EpCAM, and ABCB5 (Cheung et al. 2011). The polycomb gene product BMI1 is critically involved in the self-renewal of somatic stem cells and participates in several types of tumorigenesis. BMI1 is expressed in subpopulations of HCC stem cells and plays a role in the maintenance of TICs in HCC (Chiba et al. 2008). HCV virus-induced CSC have a molecular signature containing doublecortin and CaM kinase-like-1 (Ali et al. 2011). A stemness-related molecule in HCC is cytokeratin-19 (CK19) (Lee et al. 2012a). Expression of CK19 in HCC confers increased EMT-related protein and mRNA expression and invasive, aggressive phenotype et al. 2011; Lee et al. 2012).

#### HCC-Associated Stem Cells: Self-Renewing Cells with Various Biological Functions

HCC-associated stem cells promote tumor maintenance and progression.

HCCs can possess various self-renewing tumorigenic cell types, including CSCs sensu strictiori. These cells act as tumor-initiating cells (TICs) and tumor-propagating cells that express numerous morphologic and phenotypic features and distinct gene expression profiles (Colombo et al. 2011). CSCs of HCC express typical hepatic stem cell markers, such as CD133 and CD90, and their priming manifests in the expression of HCC lineage markers, as CD90(+) HCC-CSCs can also express glypican-3 (Ho et al. 2012). HCC-associated CSCs or stem-like cells are present in the tumors themselves, but they also occur in increased numbers in chronic liver disease, e.g., HCV-related liver cirrhosis (Behnke et al. 2013). A distinct property characterizing HCC stem cells is that they can initiate tumor growth in syngeneic or immunocompromised recipients. It is, however, difficult to judge whether such tumor-initiating cells are true CSC or non-CSC tumor cells that can give rise to growing tumors after cell transplantation (review: Sell and Leffert 2008). HCC-associated CSCs differ in several respects from other stem cells. In mouse HCC models, HCC progenitor cells give rise to fully developed cancer only when introduced into a liver showing chronic damage and compensatory hepatocyte regeneration, and tumor growth depends on cytostimulation, e.g., IL-6 (He et al. 2013). CSCs occurring in HCC affect and modulate several biological functions, including maintenance of clonogenic cells with low proliferative activity but longevity, induction of distinct growth features (also in vitro; sphere formation, colony formation, anchorage-independent growth), and tumor spread and progression. There is also growing evidence that CSC are involved in the process of hepatocarcinogenesis itself, hepatic progenitor cells exposed to carcinogens progressively accumulating genetic alterations and genomic instability, in part acquiring the features of tumor-initiating cells, and finally showing a stably transformed phenotype (Xu et al. 2010; Kim et al. 2014; Machida et al. 2015). CSCs make part of a hierarchic cancer stem cell model for solid tumors such as HCC (review: Tong et al. 2011).

HCC-associated CSCs are not only involved in the initiation and maintenance of a given tumor but are also operational in HCC recurrence (HCC recurrence-associated stem cells, RASCs), the programming and establishment of metastatic niches (niche-associated stem cells, NASCs), and growth of metastases (metastasis-associated stem cells, MASCs). The pathways leading from the expression of stem cell markers to specific cancer cell behaviors are only partially elucidated. For example, expression of the stem cell marker, CD133, in HCC cells is associated with high capacity for tumorigenicity (Suetsugu et al. 2006; Yin et al. 2007) and confers an aggressive and invasive phenotype to HCC cells through the expression of metalloproteinase-1 (MMP-1) disintegrin and metalloproteinase and (ADAM9; Kohga et al. (2010). Cytoplasmic expression of CD133 is an important risk factor for overall survival in HCC, specifically for stage III and IVA lesions (Sasaki et al. 2010).

HCC stem cells program epithelial-mesenchymal transition (EMT), a process critical for invasion and spread.

Epithelial-mesenchymal transition (EMT) is a process that occurs in many epithelial lineages and tumors derived thereof, characterized by the acquisition of a mesenchymal phenotype. The reverse process, mesenchymal-epithelial transition (MET) is less well known. EMT is driven by a distinct set of non-tissue-specific master transcriptional regulators (review: Cicchini et al. 2014). HCC stem cells are involved in the programming of EMT, a phenomenon which is crucial to invasion and spread of cancer cells and their interactions with the tissue microenvironment. On the other hand EMT creates a niche in which CSCs can settle and undergo differentiation. In this niche, with its features as an inflammatory microenvironment, mesenchymal stem cells can settle and accelerate HCC metastasis via induction of EMT (Jing et al. 2012). In cancers, EMT is chiefly characterized by disruption of intercellular contacts and connections leading to individualization of cells, enhancement of cell motility, and achievement of a primitive cell type capable to integrate into novel microenvironments (Guarino et al. 2007). In human HCC stemlike cells, the EMT promoter Twist2 induces selfrenewal of these cells in a CD24-dependent manner (Liu et al. 2014). An EMT phenotype is induced in hepatic oval cells by downregulation of microRNA-200a via targeting the beta-catenin signaling pathway (Liu et al. 2013b), which in turn affects the expression of cadherin, an important modulator of EMT. MicroRNA-200 upregulates vasohibin2 which in turn induced EMT and promotes HCC transformation (Xue et al. 2014). HCC cells that express the stem cell marker CD133 more invasive characteristics upregulation of invasion-associated genes and EMT-associated genes (Na et al. 2011). The relationship between EMT and a more invasive phenotype of cancers is linked to a distinct E-cadherin adhesion molecule repressor interactome in EMT (Hugo et al. 2011). Loss of E-cadherin, a typical feature of EMT, is associated with the acquisition of stem cell signatures and of cell with increased invasiveness (Nakagawa et al. 2014). NANOG, a major transcription factor essential for stem cell self-renewal, regulates self-renewal of CSC through the insulin-like growth factor pathway in human HCC (Shan et al. 2012) and promotes HCC invasion by inducing MEM, via activation of NODAL and CRIPTO-1 to promote SMAD3 phosphorylation and SNAIL expression. Interestingly, NANOG is preferentially expressed at the tumor edge, where invasion is most active (Sun et al. 2013a), suggesting that tumor stem cells may not be randomly distributed in liver cancer, but may rather populate distinct tumor areas. EMT can also be induced by hepatocyte growth factor secreted by hepatic stellate cells (HSCs), associated with upregulation of cancer stem cell-like properties in HCC cells (Yu et al. 2013a). The function of stem cell in the setting of EMT is modulated by tumor-associated macrophages (TAMs). In HCCs, TAMs promote cancer stem cell-like properties through TGF-beta1-induced EMT (Fan et al. 2014).

# Stem Cells in HCC: Significance for Invasion, Spread, Stroma Formation, and Angiogenesis

A meta-analysis of HCC demonstrated that the presence of cancer stem cells was significantly associated with poor histological grade and elevated serum AFP levels, but there was no correlation between the presence of cancer stem cells and tumor size, tumor stage, or chronic liver disease/cirrhosis. Cancer stem cells in HCC were significantly associated with poor survival, including overall survival and disease-free survival, suggesting that the analysis of stem cells has an impact on the prognostication of HCC biology (Ma et al. 2013). HCC harboring CD133+ cells showed an aggressive, invasive phenotype, complicated by bile duct tumor invasion and thrombi (Yu et al. 2011). The reason why CSCs in HCC can significantly affect tumor biology and prognosis is related to several CSC-modulated mechanisms involved in tumor growth, invasion spread, and generation of a tumor-specific microenvironment. CSCs interact with and modulate the function of bone marrow-derived stromal stem (BMSCs), stem cells that are recruited to form tumor stroma and that are prone to undergo progressive changes similar to that of their cancerous partner cells, including telomere attrition due to telomerase failure (review: Saeed and Iqtedar 2013). CSCs contribute to neovascularization and angiogenesis through the production transdifferentiation of proangiogenic factors, CSCs into endothelial and myoid cells, and formation of non-endothelium-lined channels called vasculogenic mimicry (reviews: Yu et al. 2010a, b; Barajas et al. 2007; Ping and Bian 2011; Zhu et al. 2012a, b). High expression levels of hepatic stem cell markers are related to tumor angiogenesis and poor prognosis in HCC. HCC with higher levels of CSC displayed higher levels of vascular endothelial growth factor and had an angiogenic pattern characteristic for aggressive tumors (Ho et al. 2006; Yang et al. 2010). Part of angiogenic effects exerted by hepatic CSC are linked to angiogenic effects of CD133 (Akita et al. 2013a). Expression of CD133 in liver tumor-initiating cells (TIC) promotes tumor angiogenesis, growth, and cell renewal via a neurotensin / interleukin-8/CXCL1 signaling pathway, causing activation of p-ERK1/2 and RAF-1, components of the mitogen-activated protein kinase/MAPK pathway (Tang et al. 2012). A second stem cell factor that affects invasive properties of HCC cells is EpCAM. There is evidence that HCC growth

and invasiveness is dependent on a subset of EpCAM+ stem cells (Terris et al. 2010). In addition to the proangiogenic effect of CSCs, neovascularization /angiogenesis in malignant neoplasms is also accomplished by homing of angiogenic circulating progenitor cells to tumors. These cells are either angiogenic stem cells already primed for endothelial differentiation or homing MSCs that will upon their settling in the tumor be fated to become angiogenic cells (review: Melero-Martin and Dudley 2011). On the other hand, endothelial progenitor cells, through their angiogenic modulation of the tumor niche, affect HCC progression. These cells promote intrahepatic metastasis via monocyte chemotactic protein-1 induction of microRNA-21 (Shih et al. 2014).

## Cancer Stem Cells and Vasculogenic Mimicry

Vasculogenic mimicry (VM) was described in 1999 based on the detection of vascular channel formation by human melanoma cells (Maniotis et al. 1999). The term, VM, denotes a phenotype resembling the pattern of embryonic vasculogenic networks and describes the plasticity of cancer cells and their precursors forming de novo vessel-like channels lined by cancerous cells, whereby these channels contribute to perfusion and transport fluid from leaky vessels (Frenkel et al. 2008). Like this, the pseudovascular structures in VM contribute to neovascularization of rapidly growing malignancies in a mode that is independent of classical angiogenesis and vasculogenesis. The development of fluid-transporting spaces in VM facilitates metastatic spread (reviews: Folberg et al. 2000; Folberg and Maniotis 2004; Ping and Bian 2011; Kirschmann et al. 2012; Liu et al. 2012; Seftor et al. 2012). In cancers, two main patterns of VM are recognized, i.e., tubular VM and patterned matrix VM. In tubular VM, fluid-containing spaces are surrounded by tumor cells, while in patterned matrix VM laminin-reactive loops with PAS-positive laminas surrounding packets of tumor cells are seen (laminin-rich looping VM). Mainly in patterned matrix VM epithelialmesenchymal transition (EMT) induced by Twist1 plays a significant pathogenic role whereby cancer cells can acquire a mesenchyme-like phenotype and may even become endothelial-like cells. In EMT-induced VM, CSCs are important actors, whereby EMT itself is potent for the acquisition and maintenance of stem-like features of cancer cells (Fan et al. 2013). Twist1 involved in EMT can stimulate cancer cell migration (Matsuo et al. 2009) and via this effect facilitates VM which requires tumor cell movements. In hypoxiainduced VM, Bcl-2 can induce a major endothelial adhesion molecule in tumor cells, i.e., VE-cadherin (Zhao et al. 2012). However, VM is resistant to antiangiogenic therapies (Fan and Sun 2010). Apart from factors regulating EMT, several other molecules are active in the promotion of VM, including the PI3-K signaling pathway, matrix metalloproteinases, laminin-5-gamma chain, protein tyrosine kinases, epithelial cell kinase (ECK), focal adhesion kinase, Eph receptors, Nodal, Notch4, hypoxia-inducible factor, galectin-3, bone morphogenetic proteins, cAMP signaling, cyclooxygenase-2, inhibitor of DNA binding 2/Id2, autophagy-related proteins, and VEGF (Hess et al. 2001; Hess et al. 2007; Sun et al. 2007; Lissitzky et al. 2009; Su et al. 2009; Fan and Sun 2010; Hardy et al. 2010; McAllister et al. 2010; Paulis et al. 2010; Che et al. 2011; Han et al. 2011a; Sun et al. 2012; Vartanian et al. 2013; Ding et al. 2014).

Both tubular and patterned matrix VM occur in HCC and is correlated with aggressiveness and poor clinical prognosis (Sun et al. 2006; Zhao et al. 2006; Guzman et al. 2007). Patterned matrix VM in HCC was associated with larger tumor, vascular invasion, high tumor grade, and late stage (Liu et al. 2011b). Similar to other cancers, VM induced by HCC cells is related to the induction of EMT, in that HCC cells in VM express the EMT inducer, Twist1 (Sun et al. 2010). Via EMT, Twist1 directly induces VM in hypoxic HCC cells (Ma et al. 2011) and promotes a metastatic phenotype (Lee et al. 2006). EMT-associated VM induced by Twist1 is stimulated by Bcl-2, which binds to and activates Twist1 under hypoxic conditions (Sun et al. 2011). Various factors present in EMT microenvironments are involved

VM. The extracellular protease ADAMTS1 is a critical enzyme for cancer cells to acquire endothelial-like properties (Casal et al. 2010), and VM in HCCs is associated with the expression of osteopontin and matrix metalloproteinase-2 (Liu et al. 2011d). In the course of EMT-induced VM, poorly differentiated HCC cells can become CD31 positive and therefore phenotypically resemble endothelial cells (Zhao et al. 2007).

HCC-associated CSCs play a role VM. In the process of VM, a stem cell-like phenotype is involved, whereby cells forming the vascular channels express stemness markers, including CD133 (Yao et al. 2011; Lai et al. 2012; Valyi-Nagy et al. 2012). VM channel formation in HCCs is associated with EMT, dedifferentiation of tumor cells, and increased expression of stemness genes (Lirdprapamongkol et al. 2012). Stem cell-induced VM is particularly prevalent in EMT that develops in hypoxia. Expression of Slug, a potent inducer of EMT, in HCCs promotes the biogenesis and maintenance of a CSC subpopulation that is capable to produce VM (Sun et al. 2013b). EMT itself promotes the formation of HCC cells with stem-like properties via Twist1-Bmi1 signaling, Bmi1 belonging to the polycomb repressive complex 1 (Wu and Yang 2011), and this connection may favor EMT-induced VM mediated by cancer stem cells.

# CSCs in Tumor Recurrence and Metastatic Spread: RASCs, NASCs, and MASCs

Tumor recurrence is traditionally viewed as a process that originates from remaining cancer cells that have not been radically removed and/or were resistant to chemo- or radiotherapy. However, within the population of recurrent HCC cells, cells with stemlike features were identified (Yi and Nan 2008; Xu et al. 2010a, b), and it is assumed that such cells contribute to recurrence and significantly modify the recurrence phenotype and pattern. For hepatic recurrence, recurrence-associated stem cells (RASCs) may either represent (a) surviving HCC-CSCs capable to resume growth following therapy, (b) novel stem cell species arising from the recurrent tumor, or (c) stem cells derived from

an extrahepatic site and homing to the recurrent tumor with its new microenvironment. Tumor stem cells and their fully malignant offspring play a critical role in the biogenesis of metastatic niches. In human HCCs, expression of the two stem cell markers CD90 and EpCAM was higher in the early recurrence group (Guo et al. 2014). As niche-associated stem cells (NASCs), CSCs undergo a complex crosstalk with normal cells in the prometastatic niche, in that they secrete factors that affect stromagenesis, recruitment of tumor-associated macrophages (TAMs) and tumor-associated endothelial cells (TECs), and angiogenesis (Gupta et al. 2014). These factors include the TGF-beta/ Smad system, PDGF, VEGF, and IL-6. Similar to the complex function of the portal fibroblast, which provided a niche for controlled cholangiocyte turnover (Wells 2014), prometastatic niches contain a fibroblastoid population that interacts with NASCs, with a mutual regulation of the cells involved. Part of fibroblastoid cells and angiogenic cells constituting the prometastatic niche are derived from mesenchymal stem cells (MSCs) that circulate in the blood and home to the tumor microenvironment (Wang and Chen 2013), where they interact with tumor cells and NASCs. The mode of this interaction is not yet well known, but MSCs rarely fuse with HCC cells in murine models (Li et al. 2013). HCC cells and their stromal component themselves modulate the migratory behavior of MSCs (Garcia et al. 2011). As the tumor microenvironment is often hypoxic and/or deficient in nutrients, nicheassociated tumor cells have developed strategies to circumvent this stress situation. CD133(+) HCC-CSCs showed higher survival, less apoptosis, and higher clonogenicity under hypoxic and nutrient deprivation stress, in part associated with increased autophagy (Song et al. 2013). CD133 is involved in the regulation of autophagy and glucose uptake (Chen et al. 2013d). Following their exit from a primary or recurrent tumor, or a metastatic niche, RASCS and NASCs enter circulation and can become metastasis-associated stem cells or MASCs. There is evidence that circulating HCC stem cell-like cells capable to generate metastatic recurrence (MASCs) express certain markers that allow their identification, such as the stem cell marker CD90 (Yamashita et al. 2013), the oncofetal marker Lin28B (Cheng et al. 2013), and the G protein-coupled receptor 87/GPR87 (Yan et al. 2013). Expression of CD133 in liver CSC plays a critical role in hematogenous metastasis of HCC (Hou et al. 2012; Chow et al. 2013). CD133positive CSCs play a role in repeated pulmonary recurrence of HCC (Toshima et al. 2013). In regard to prognostication, the various HCC stem cell markers differ in several respects. While expression of CD133 was more frequently found in small tumors in cirrhotic livers and early stage disease, expression of EpCAM characterized small tumors and poor differentiation and was an independent prognosticator at all stages (Chan et al. 2014). Can circulating CSCs rehome to primary cancers or metastatic niches and modulate the features of their tumor of origin?

### **Tumor Stem Cells in Nontumorous Liver Tissue**

CSC are readily detectable in HCC tissue itself by means of their distinct marker profile, but cells with similar or the same features also occur in the nonneoplastic parenchyma of livers harboring HCC. Cell lines producing colonies in soft agar and showing anchorage-independent growth were isolated form liver resections with HCC, and these cells revealed the marker profile of CSC (Zhang et al. 2010).

# Cells with Stem Cell-Like Features in Special Variants of Hepatocellular Carcinoma and in Precursor Lesions

Tumor stem cells seem to be involved in pathogenesis of scirrhous HCC. In this variant of HCC, small neoplastic cells were reactive for cytokeratin 7 and ATP-binding cassette transporter G2. Part of these cells (termed type 1 cells) stained for cytokeratin 19, neural cell adhesion molecule (NCAM), and epithelial cell adhesion molecule, while another subset (termed type 2 cells) did not show this immunophenotype. In type 3 scirrhous HCC, no small tumor cells with stem cell features were present. It was suggested that type 1 and type

2 cells in scirrhous HCC correspond to a side population of cultured cells (Fujii et al. 2008). Apart from conventional HCC, features of stemness are also present in dysplastic hepatocytes (Lingala et al. 2010), fibrolamellar hepatocellular carcinoma (Zenali et al. 2010), and hepatoblastomas (Akita et al. 2013).

#### **Stem Cells in Hepatoblastomas**

Hepatoblastomas (HBs) have been shown to contain cells with stem-like features. Hepatoblasts, a putative normal counterpart of HB, show a pattern of proteins shared by hepatic stem cells, i.e., CK19, human epithelial antigen-125/HEA-125, and OV-6 (Crosby et al. 1998). This has in particular been observed in HBs of low differentiation, such as small cell undifferentiated HB (see the respective chapter). In subsets of HB, expression of CK19, Oct-3/4, and EpCAM was observed (Yun et al. 2013). HB cells can express proteins that are present in human fetal liver, including delta-like protein/DLK, a membrane protein of unknown function (Dezso et al. 2008; Yun et al. 2013).

### Biogenesis of HCC-Associated Stem Cells

## Mechanisms of Stem Cell Formation in the Liver and Liver Neoplasms

Recent findings showed that several mechanisms can elicit tumor stem cells and stem cell signatures in normal liver, regenerating liver, and HCCs. They comprise the expression of transcription factors (Sox2, c-Myc, LIN28, NANOG), several signaling pathways (TGF-beta, Wnt/beta-catenin signaling), factors modulating the cancer microenvironment (Notch signaling), microRNAs, and differential DNA methylation patterns (reviews: Zheng et al. 2013; Bogaerts et al. 2014; Dang et al. 2014). Expression of transforming growth factor beta play an important role in signaling pathways taking place in hepatic and intestinal stem cell niches (review: Majumdar et al. 2012).

Chronic and constant TGF-beta stimulation results in the generation of hepatoma-initiating cells derived from hepatic progenitor cells (Wu et al. 2012), and TGF-beta secreted by HCC cells mediates the crosstalk between cancer cells and the tumor microenvironment (Gupta et al. 2014).

How are the distinct growth and survival features of CSCs regulated, in particular their critical capacity for self-renewal? Overall, the expression of stemness-related proteins by HCC cells is related to increased telomere lengths, increased expression of hTERT and shelterin complex proteins, and increased chromosomal/genomic instability in comparison with conventional HCC (Idrissi et al. 2013). Maintenance of telomeres is compromised in HCCs, and it is expected that a derangement of the telomere platform may be an early event in hepatocarcinogenesis, eventually already involving CSCs (Ye et al. 2010). Specifically, there is evidence that abnormal stem cells may lack telomerase, an enzyme responsible for telomere extension (reviews: Gardano et al. 2013; Saeed and Iqtedar 2013). Telomeres are the terminal elements at the ends of linear chromosomes and are critically involved in the prevention of exonucleolytic degradation, interand intrachromosomal fusion, and subsequent chromosomal instability (Ye et al. 2010; Calado and Dumitriu 2013). Telomere integrity is critically required for the unlimited replicative potential of cancer cells. The stability of telomeres depends on the binding of the six-protein subunit complex shelterin to telomeric repetitive DNA sequences (Palm and de Lange 2008; Xin et al. 2008). Further proteins playing a role in telomere stability are the sirtuins (Rodriguez et al. 2013). Telomerase elongates telomeres and is recruited by homeobox telomere-bending protein1/HOT1 (Kappel et al. 2013). Furthermore, the subunit telomerase reverse transcriptase (TERT) has additional functions, including transcriptional regulation and metabolic reprogramming (Low and Tergaonkar 2013). All telomeric proteins involved in telomere function form, together with the telomere proper, the telosome (Liu et al. 2004; Folini et al. 2009).

Epigenetic mechanisms play a central role in the generation of hepatic stem cells and side population cells occurring in HCC. Side population cells of HCC have a high rate of DNA hypermethylation as compared with their corresponding non-side population cells, and the differentially methylated genes in side population cells were involved in numerous signaling pathways (Zhai et al. 2013). Among stem cell molecules regulated by epigenetic mechanisms is CD133, this regulation pathway involving TGF-beta1 which inhibits DNA methyltransferase 1 and DNA methyltransferase 3beta expression and subsequent demethylation of promoter-1 (You et al. 2010). CD133 expression is also promoted by hypoxia, in that hypoxiainducible factors activate CD133 promoter via family transcription factors (Ohnishi et al. 2013). Certain transcription factors and transcription repressors are preferentially active in stem and progenitor cells and affect epigenetic mechanisms. BORIS (brother of regulator of imprinted sites) is a protein encoded by the CTCFL gene, is a paralog of the transcription suppressor CTCF (Tiffen et al. 2013), and is an ubiquitous 11-zinc finger protein with highly versatile functions, including transcriptional silencing and organization of epigenetically controlled chromatin insulators that regulate imprinted genes in the soma both in normal and cancerous tissues (Klenova et al. 2002; Loukinov et al. 2002; Rosa-Garrido et al. 2012). As BORIS expression in HCC is correlated with expression of the stem cell marker CD90, BORIS may itself be a property of hepatic and HCC-associated stem cells (Chen et al. 2013b).

## Pathways from Primed Stem Cells to Differentiated Cells

In parallel to normal stem cells it may be anticipated that CSC can undergo priming to more differentiated neoplastic cells. This is an important study field, because it potentially opens pathways for novel treatment strategies. One factor involved in the switch turning CSC to differentiated cells is the Hippo pathway. Hippo signaling

has a dual regulation of Hippo in liver tumor suppression as well as a transition of oval stem cells to fully differentiated hepatocytes and hepatocyte-like cells in tumors (Zheng et al. 2011). A second crucial pathway involved in the phenotypic shift of CSC to a more differentiated offspring is the WNT/beta-catenin-signaling pathway (Yang and Poon 2008). Expression of beta-catenin is pertinent in hepatic oval cell activation and differentiation (Nejak-Bowen and Monga 2011), and it is expected that the same mechanism is operational in HCC and related neoplasms. As discussed in the respective chapter, immature cells forming undifferentiated clusters in hepatoblastomas are markedly reactive for beta-catenin and seem to give rise to more differentiated but rapidly proliferating offspring surrounding the stem-like cell clusters (Zimmermann 2005).

## Regulation of Stem Cells by microRNAs and Other Noncoding RNAs

The function of HCC-associated stem cells is itself regulated by several factors and signaling pathways, including microRNAs (reviews: Chai and Ma 2013; Qi et al. 2013; Chen and Verfaillie 2014). In HCCs and HCC cell lines several and in part specific microRNAs have been identified (Xu et al. 2013). miRNA-130b was shown to promote liver tumor-initiating cell growth and self-renewal through induction of nuclear protein 1 by tumor p53 (Ma et al. 2010). In isolated subsets of side population cells, silencing of microRNA-21 reduced migration and invasion of cancer cells, while overexpression of mRNA-21 drastically inhibited the expression of PTEN, RECK, and PDCD4 proteins (Zhou et al. 2013). MicroRNA-142-3p regulates the expression of the functional hepatic cancer stem cell marker CD133 (Chai et al. 2014). microRNA-181 is highly expressed in embryonic liver tissue and in isolated hepatic stem cells. Specifically, inhibition of microRNA-181 resulted in reduction in EpCAM+ HCC cells and tumor-initiating capacity (Ji et al. 2009). microRNA-150 inhibits human CD133+ liver cancer stem cells through negative

regulation of the transcription factor cMyb, associated with downregulation of the cell cycle regulator cyclin D1 and the cell survival regulator Bcl-2 (Zhang et al. 2012). micro-612 suppresses the stemness of HCC CSCs via the Wnt/betacatenin-signaling pathway (Tang et al. 2014). The tumorigenicity of cancer stem-like cells derived from HCC is regulated by microRNA-145, probably through modulation of the downstream target, the stem cell-related gene Oct4 (Jia et al. 2012; Wang et al. 2013). microRNA-148a inhibits EMT and several CSC properties and by this impairs metastasis of HCC (Yan et al. 2013). Lin-41, a stem cell-specific E3 ligase and a target of the tumor suppressor microRNA let-7 mediates ubiquitinylation and degradation of microRNA pathway protein Ago2. Lin-41 is overexpressed in HCC, where it is involved in growth control through regulating the RISC complex proteins Ago1 and Ago2 to inhibit microRNA-mediated gene silencing and promoting the expression of oncogenic proteins (Chen et al. 2013a). Lin-41 expression in HCC confers an aggressive phenotype associated with early recurrence and poor survival, suggesting that Li41 is involved in HCC CSC. Hepatic stem cell biology is also affected by long noncoding RNAs. Oncofetal long noncoding RNA PVT1 promotes proliferation and stem cell-like properties of HCC cells by stabilizing NOP2 (Wang et al. 2014).

#### Modulation of Stem Cell Biogenesis and Function by Microvesicles and Exosomes: Can Cancer Stem Cells "Infect" Other Cells and Transmit Them a Malignant Phenotype?

Mesenchymal stem cells (MSC) as a potential source of hepatic cells possess a complex stem cell secretome that can be transferred to other cell systems through exosomal pathways (Bruno et al. 2013; Kupcova 2013; Xiong et al. 2013). Microvesicles of MSC can serve as trophic shuttles for diverse cell types, including stem cells (Aliotta et al. 2012; Mokarizadeh et al. 2013) and act as a complex paracrine information system within cell interactomes (Quesenberry and

Aliotta 2010), e.g., in regeneration and tissue repair (Camussi et al. 2010; Anthony and Shiels and the induction of epithelialmesenchymal transition (Garnier et al. 2013). Via this mechanism, stem cell properties are maintained, while loss of the pathway may result in differentiation of stem cells (Bauer et al. 2011). Specifically, microvesicles released from stem cells confer a stem cell-like phenotype to other stem cells and damaged cells, evoking stem cell maintenance and self-regenerative programs (Herrera et al. 2010; Quesenberry et al. 2012; Camussi et al. 2013), but they may also transfer stem cell-like features to neoplastic cells being in close contact with them (Muralidharan-Chari et al. 2010; Lee et al. 2011; Pap 2011).

A critical property of stem cell microvesicles and exosomes is their genetic information cargo, i.e., they are loaded with microRNAs and mRNAs, genetic information that can be transferred among cells through microvesicle release (Yuan et al. 2009; Deregibus et al. 2010; Lee et al. 2012b). The molecular mechanisms by which genetic information such as microRNAs are loaded into microvesicles/exosomes are not yet well known, but distinct RNA-related zipcode-like sequences seem to act as sentinel sequences for these processes (Bolukbasi et al. 2012). Multipotent mesenchymal stem cells can affect and modulate their partners and other cells through CD133/prominin-1, contained in lipid rafts of these cells, by release of microvesicles. Similarly, cancer cells possess exosomes containing CD133 and numerous other proteins and microRNAs that are involved in metastatic pathways (Rappa et al. 2013). Human liver stem cell-derived microvesicles inhibit HCC growth in immunodeficient mice by releasing antitumor microRNAs (Fonsato et al. 2012). Informational cargo transported and transmitted by exosomes also modulates processes that are associated with and required by stem cell-driven cancer growth and spread. Endothelial progenitor cells located within tumors produce microvesicles that activate an angiogenic program in endothelial cells by horizontal transfer of specific mRNAs (Deregibus et al. 2007).

# Systemic and Hepatic Mesenchymal Stem Cells and Stem-Like Cells: Modulation of HCC Biology

In murine models, there is evidence that HCC may originate from genetically mutated MSCs (Gong et al. 2013). MSC interact with HCC cells and modulate their behavior and affect interactions of HCC cells with stromal and vascular cells. In the nude mouse tumor transplantation model, BMSCs could promote the growth of small vessels and increased microvascular density (Gong et al. 2013a, b; Kupcova 2013), and in another murine HCC model, hematogenic MSC could modulate their microenvironment via secreted cytokines that promoted tumor initiation, growth, and homing to tumor sites (Gong et al. 2013). Hepatic stellate cells (HSC) and their precursors are specific mesenchymal cells of the liver that exert a profound impact on growth, differentiation, and morphogenesis of several other hepatic cell systems and neoplasms derived thereof (review: Yin et al. 2013).

#### **Dormant Cancer Stem Cells**

Dormant cancer stem cells (D-CSC) are defined as cells that remain intact after apparent eradication of cancer, including HCC, and are capable to maintain a transformed phenotype and to generate recurrent cancer. In one study, such D-CSC were detected in human HCC cell population grown in immunodeficient murine hosts after chemotherapy in the form of human AFP+/CD13+/PCNA cells (Martin-Padura et al. 2012).

#### References

Akita M, Tanaka K, Matsumoto S, Komatsu K, Fujita K (2013a) Detection of the hematopoietic stem and progenitor cell marker CD133 during angiogenesis in three-dimensional collagen gel culture. Stem Cells Int 2013:10

Akita M, Tanaka K, Murai N, Matsumoto S, Fujita K, Takaki T, Nishiyama H (2013b) Detection of CD133 (prominin-1) in a human hepatoblastoma cell line (HuH-6 clone 5). Microsc Res Tech 76:844–852

References 17

- Ali N, Allam H, May R, Sureban SM, Bronze MS, Bader T, Umar S, Anant S, Houchen CW (2011) Hepatitis C virus-induced cancer stem cell-like signatures in cell culture and murine tumor xenografts. J Virol 85:12292–12303
- Aliotta JM, Lee D, Puente N, Faradyan S, Sears EH, Amaral A, Goldberg L, Dooner MS et al (2012) Progenitor/stem cell fate determination: interactive dynamics of cell cycle and microvesicles. Stem Cells 21:1627–1638
- Alisi A, Baffet G (2012) Self-renewal of tumor-initiating cells: what's new about hepatocellular carcinoma? Gastroenterology 142:1414–1416
- Alison MR, Poulsom R, Forbes SJ (2001) Update on hepatic stem cells. Liver 21:367–373
- Alison MR, Lovell MJ (2005) Liver cancer: the role of stem cells. Cell Prolif 38:407–421
- Alison MR, Islam S, Lim S (2009) Stem cells in liver regeneration, fibrosis and cancer: the good, the bad and the ugly. J Pathol 217:282–298
- Anderson LH, Boulanger CA, Smith GH, Carmeliet P, Watson CJ (2011) Stem cell marker prominin-1 regulates branching morphogenesis, but not regenerative capacity, in the mammary gland. Dev Dyn 240:674–681
- Anthony DF, Shiels PG (2013) Exploiting paracrine mechanisms of tissue regeneration to repair damaged organs.

  Transplant Res 2(1):10
- Barajas M, Franchi F, Clavel C, Aranguren XL, Kramer MG, Abizanda G, Merino J et al (2007) Multipotent adult progenitor cells (MAPC) contribute to hepatocarcinoma neovasculature. Biochem Biophys Res Commun 364:92–99
- Bauer N, Wilsch-Bräuninger M, Karbanova J, Fonseca AV, Strauss D, Freund D, Thiele C et al (2011) Haematopoietic stem cell differentiation promotes the release of prominin-1/containing membrane vesicles-a role of the endocytic-exocytic pathway. EMBO Mol Med 3:398–409
- Becker G, Schmitt-Graeff A, Ertelt V, Blum HE, Allgaier HP (2007) CD177 (c-kit) expression in human hepatocellular carcinoma. Clin Oncol (R Coll Radiol) 19:204–208
- Behnke MK, Reimers M, Fisher RA (2013) Stem cell and hepatocyte proliferation in hepatitis C cirrhosis and hepatocellular carcinoma: transplant implications. Ann Hepatol 13:45–53
- Bird TG, Lorenzini S, Forbes SJ (2008) Activation of stem cells in hepatic diseases. Cell Tissue Res 331:283–300
- Blechacz B, Mishra L (2013) Hepatocellular carcinoma biology. Recent Results Cancer Res 190:1–20
- Bogaerts E, Heindryckx F, Vandewynckel YP, Van Grunsven LA, Van Vlierberghe H (2014) The roles of transforming growth factor-b, Wnt. Notch and hypoxia on liver progenitor cells in primary liver tumours (review). Int J Oncol 44:1015–1022
- Bolukbasi MF, Mizrak A, Ozdener GB, Madlener S, Ströbel T, Erkan EP, Fan JB et al (2012) miR-1289 and "zipcode"-like sequence enrich mRNAs in microvesicles. Mol Ther Nucleic Acids 1:e10

Bourseau-Guilmain E, Griveau A, Benoit JP, Garcion E (2011) The importance of the stem cell marker prominin-1/CD133 in the uptake of transferrin and in iron metabolism in human colon cancer Caco-2 cells. PLoS One 6, e25515

- Bruno S, Collino F, Deregibus MC, Grange C, Tetta C, Camussi G (2013) Microvesicles derived from human bone marrow mesenchymal stem cells inhibit tumor growth. Stem Cells Dev 22:758–771
- Calado RT, Dumitriu B (2013) Telomere dynamics in mice and humans. Semin Hematol 50:165–174
- Camussi G, Deregibus MC, Bruno S, Cantaluppi V, Biancone L (2010) Exosomes/microvesicles as a mechanism of cell-to-cell communication. Kidney Int 78:838–848
- Camussi G, Deregibus MC, Cantaluppi V (2013) Role of stem-cell-derived microvesicles in the paracrine action of stem cells. Biochem Soc Trans 41:283–287
- Casal C, Torres-Collado AX, Plaza-Calonge MC, Martino-Echarri E, Ramon Y, Cajal S et al (2010) ADAMTS1 contributes to the acquisition of an endothelial-like phenotype in plastic tumor cells. Cancer Res 70:4676–4686
- Chai S, Ma S (2013) Clinical implications of miRNAs in liver cancer stem cells. Chin J Cancer 32:419–426
- Chai S, Tong M, Ng KY, Kwan PS, Chan YP, Fung TM, Lee TK, Wong N, Xie D, Yuan YF, Guan XY et al (2014) Regulatory role of miR-142-3p on the functional hepatic cancer stem cell marker CD133. Oncotarget 5:5725–5735
- Chan AW, Tong JH, Chan SL, Lai PB, To KF (2014) Expression of stemness markers (CD133 ad EpCAM) in prognostication of hepatocellular carcinoma. Histopathology 64:935–950
- Che N, Zhao XL, Sun T, Zhao XM, Gu Q, Dong XY, Zhao N, Liu YR, Yao Z, Sun BC (2011) The role of Twist1 in hepatocellular carcinoma angiogenesis: a clinical study. Hum Pathol 42:840–847
- Chen Y, Yu D, Zhang H, He H, Zhang C, Zhao W, Shao RG (2012) CD133(+)EpCAM(+) phenotype possesses more characteristics of tumor initiating cells in hepatocellular carcinoma Huh7 cells. Int J Biol Sci 8:992–1004
- Chen YL, Yuan RH, Yang WC, Hsu HC, Jeng YM (2013a) The stem cell E3-ligase Lin-41 promotes liver cancer progression through inhibition of microRNA-mediated gene silencing. J Pathol 229:486–496
- Chen K, Huang W, Huang B, Wie Y, Li B, Ge Y, Qin Y (2013b) BORIS, brother of the regulator of imprinted sites, is aberrantly expressed in hepatocellular carcinoma. Genet Test Mol Biomarkers 17:160–165
- Chen Q, Khoury M, Limmon G, Choolani M, Chan JK, Chen J (2013c) Human fetal hepatic progenitor cells are distinct from, but closely related to, hematopoietic stem/progenitor cells. Stem Cells 31:1160–1169
- Chen H, Luo Z, Dong L, Tan Y, Yang J, Feng G, Wu M, Li Z, Wang H (2013d) Cd133+/prominin-1-mediated autophagy and glucose uptake beneficial for hepatoma cell survival. PLoS One 8, e56878

- Chen Y, Verfaillie CM (2014) MicroRNAs: the fine modulators of liver development and function. Liver Int 34(7):976–990. doi:10.1111/liv.12496
- Cheng SW, Tsai HW, Lin YJ, Cheng PN, Chang YC, Yen CJ, Huang HP, Chuang YP et al (2013) Lin28B is an oncofetal circulating cancer stem cell-like marker associated with recurrence of hepatocellular carcinoma. PLoS One 8, e80053
- Cheung PF, Cheng CK, Wong NC, Ho JC, Yip CW, Liu VC, Cheung AN, Fan ST et al (2011) Granulin-epithelin precursor is an oncofetal protein defining hepatic cancer stem cells. PLoS One 6, e28246
- Chiang CH, Huo TI, Sun CC, Hsieh JH, Chien Y, Lu KH, Lee SD (2013) Induced pluripotent stem cells and hepatic differentiation. J Chin Med Assoc 76:599–605
- Chiba T, Kita K, Zheng YW, Yokosuka O, Saisho H, Iwama A, Nakauchi H, Taniguchi H (2006) Side population purified from hepatocellular carcinoma cells harbors cancer stem cell-like properties. Hepatology 44:240–251
- Chiba T, Miyagi S, Saraya A, Aoki R, Seki A, Morita Y, Yonemitsu Y, Yokosuka O et al (2008) The polycomb gene product BMI1 contributes to the maintenance of tumor-initiating side population cells in hepatocellular carcinoma. Cancer Res 8:7742–7749
- Chiba T, Kamiya A, Yokosuka O, Iwama A (2009) Cancer stem cells in hepatocellular carcinoma: recent progress and perspective. Cancer Lett 286:145–153
- Chow AK, Ng L, Lam CS, Wong SK, Wan TM, Cheng NS, Yau TC, Poon RT, Pang RW (2013) The enhanced metastatic potential of hepatocellular carcinoma (HCC) cells with sorafenib resistance. PLoS One 8, e78675
- Cicchini C, Amicone L, Alonzi T, Marchetti A, Mancone C, Tripodi M (2014) Molecular mechanisms controlling the phenotype and the EMT/MET dynamics of hepatocyte. Liver Int 35(2):302–310. doi:10.1111/liv.12577
- Colombo F, Baldan F, Mazzucchelli S, Martin-Padura I, Marighetti P, Cattaneo A, Foglieni B et al (2011) Evidence of distinct tumour-promoting cell populations with different properties in primary human hepatocellular carcinoma. PLoS One 6, e21369
- Crosby HA, Hubscher SG, Joplin RE, Kelly DA, Strain AJ (1998) Immunolocalization of OV-6, a putative progenitor cell marker in human fetal and diseased pediatric liver. Hepatology 28:980–985
- Dabeva MD, Shafritz DA (2003) Hepatic stem cells and liver repopulation. Semin Liver Dis 23:349–362
- Dan YY, Yeoh GC (2008) Liver stem cells: a scientific and clinical perspective. J Gastroenterol Hepatol 23:687–698
- Dang HT, Budhu A, Wang XW (2014) The origin of cancer stem cells. J Hepatol 60(6):1304–1305. doi:10.1016/j. hep.2014.03.001
- Darwiche H, Petersen BE (2010) Biology of the adult hepatic progenitor cell: "ghosts in the machine". Prog Mol Biol Transl Sci 97:229–249
- Davoodian N, Lotfi AS, Soleimani M, Mola J (2014) MicroRNA-122 overexpression promotes hepatic

- differentiation of human adipose tissue-derived stem cells. J Cell Biochem 115(9):1582–1593. doi:10.1002/jcb.24822
- Deregibus MC, Cantalupi V, Calogero R, Lo IM, Tetta C, Biancone L, Bruno S et al (2007) Endothelial progenitor cell derived microvesicles activate an angiogenic program in endothelial cells by a horizontal transfer of mRNA. Blood 110:2440–2448
- Deregibus MC, Tetta C, Camussi G (2010) The dynamic stem cell microenvironment is orchestrated by microvesicle-mediated transfer of genetic information. Histol Histopathol 25:397–404
- Dezso K, Halasz J, Bisgaard HC, Paku S, Turanyi E, Schaff Z, Nagy P (2008) Delta-like protein (DLK) is a novel immunohistochemical marker for human hepatoblastoma. Virchows Arch 452:443–448
- Ding YP, Yang XD, Wu Y, Xing CG (2014) Autophagy promotes the survival and development of tumors by participating in the formation of vasculogenic mimicry. Oncol Rep 31:2321–2327
- Dong HH, Xiang S, Liang HF, Li CH, Zhang ZW, Chen XP (2013) The niche of hepatic cancer stem cell and cancer recurrence. Med Hypotheses 80:666–668
- Fan YZ, Sun W (2010) Molecular regulation of vasculogenic mimicry in tumors and potential tumortarget therapy. World J Gastrointest Surg 2:117–127
- Fan L, He F, Liu H, Zhu J, Liu Y, Yin Z, Wang L, Guo Y, Wang Z, Yan Q, Huang G (2011) CD133: a potential indicator for differentiation and prognosis of human cholangiocarcinoma. BMC Cancer 11:320
- Fan YL, Zheng M, Tang YL, Liang XH (2013) A new perspective of vasculogenic mimicry: EMT and cancer stem cells (review). Oncol Lett 6(5):1174–1180
- Fan QM, Jing YY, Yu GF, Kou XR, Ye F, Gao L, Li R, Zhao QD, Yang Y, Lu ZH, Wei LX (2014) Tumorassociated macrophages promote cancer stem cell-like properties via transforming growth factor-beta1induced epithelial-mesenchymal transition in hepatocellular carcinoma. Cancer Lett 352:160–168
- Fargeas CA, Karbanova J, Jaszai J, Corbeil D (2011) CD133 and membrane microdomains: old facets for future hypotheses. World J Gastroenterol 17:4149–4152
- Fausto N, Campbell JS (2003) The role of hepatocytes and oval cells in liver regeneration and repopulation. Mech Dev 120:117–130
- Folberg R, Hendrix MJ, Maniotis J (2000) Vasculogenic mimicry and tumor angiogenesis. Am J Pathol 156:361–381
- Folberg R, Maniotis AJ (2004) Vasculogenic mimicry. APMIS 112:508–525
- Folini M, Gandellini P, Zaffaroni N (2009) Targeting the telosome: therapeutic implications. Biochim Biophys Acta 1792:309–316
- Fonsato V, Collino F, Herrera MB, Cavallari C, Deregibus MC, Cisterna B, Bruno S et al (2012) Human liver stem cell-derived microvesicles inhibit hepatoma growth in SCID mice by delivering antitumor microRNAs. Stem Cells 30:1985–1998

References 19

Forbes S, Vig P, Poulsom R, Thomas H, Alison M (2002) Hepatic stem cells. J Pathol 197:510–518

- French SW, Vitocruz E, French BA (2013) Balloon liver cells forming Mallory-Denk-bodies are progenitor cells. Exp Mol Pathol 95:117–120
- Frenkel S, Barzel I, Levy J, Lin AY, Bartsch DU, Majumdar D, Folberg R, Pe'er J (2008) Demonstrating circulation in vasculogenic mimicry patterns of uveal melanoma by confocal indocyanine green angiography. Eye (Lond) 22:948–952
- Fujii T, Zen Y, Harada K, Niwa H, Masuda S, Kaizaki Y, Watanabe K, Kawashima A et al (2008) Participation of liver cancer stem/progenitor cells in tumorigenesis of scirrhous hepatocellular carcinoma-human and cell culture study. Hum Pathol 39:1185–1196
- Garcia MG, Bayo J, Bolontrade MF, Sganga L, Malvicini M, Alaniz L, Aquino JB, Fiore E et al (2011) Hepatocellular carcinoma cells and their fibrotic microenvironment modulate bone marrowderived mesenchymal stromal cell migration in vitro and in vivo. Mol Pharm 8:1538–1548
- Gardano L, Pucci F, Christian L, Le Bihan T, Harrington L (2013) Telomeres, a busy platform for cell signaling. Front Oncol 3:146
- Garnier D, Magnus N, Meehan B, Kislinger T, Rak J (2013) Qualitative changes in the proteome of extracellular vesicles accompanying cancer cell transition to mesenchymal state. Exp Cell Res 319:2747–2757
- Gaudio E, Carpino G, Cardinale V, Franchitto A, Onori P, Alvaro D (2009) New insights into liver stem cells. Dig Liver Dis 41:455–462
- Gong P, Wang Y, Zhang J, Wang Z (2013a) Bone marrow mesenchymal stem cells in hepatocellular carcinoma. Front Biosc 18:811–819
- Gong P, Wang Y, Wang Y, Jin S, Luo H, Zhang J, Bao H, Wang Z (2013b) Effect of bone marrow mesenchymal stem cells on hepatocellular carcinoma in microcirculation. Tumour Biol 34:2161–2168
- Grompe M (2005) Bone marrow-derived hepatocytes. Novartis Found Symp 265:20–27
- Guarino M, Rubino B, Ballabio G (2007) The role of epithelial-mesenchymal transition in cancer pathology. Pathology 39:305–318
- Guest I, Ilic Z, Ma J, Grant D, Glinsky G, Sell S (2010) Direct and indirect contribution of bone marrow derived cells to cancer. Int J Cancer 126:2308–2318
- Guo Z, Li LQ, Jiang JH, Ou C, Zeng LX, Xiang BD (2014) Cancer stem cell markers correlate with early recurrence and survival in hepatocellular carcinoma. World J Gastroenterol 20:2098–2106
- Gupta DK, Singh N, Sahu DK (2014) TGF-b mediated crosstalk between malignant hepatocyte and tumor microenvironment in hepatocellular carcinoma. Cancer Growth Metastasis 7:1–8
- Guzman G, Cotler SJ, Lin AY, Maniotis AJ, Folberg R (2007) A pilot study of vasculogenic mimicry immunohistochemical expression in hepatocellular carcinoma. Arch Pathol Lab Med 131:1776–1781

- Hagiwara S, Kudo M, Ueshima K, Chung H, Yamaguchi M, Takita M, Haji S, Kimura M et al (2011) The cancer stem cell marker CD133 is a predictor of the effectiveness of S1+ pegylated interferon a-2b therapy against advanced hepatocellular carcinoma. J Gastroenterol 46:212–221
- Han C, Sun B, Wang W, Cai W, Lou D, Sun Y, Zhao X (2011a) Overexpression of microtubule-associated protein-1 light chain 3 is associated with melanoma metastasis and vasculogenic mimicry. Tohoku J Exp Med 223:243–251
- Han S, Dziedzic N, Gadue P, Keller GM, Gouon-Evans V (2011b) An endothelial cell niche induces hepatic specification through dual repression of Wnt and Notch signaling. Stem Cells 29:217–228
- Hang HL, Xia Q (2014) Role of BMSCs in liver regeneration and metastasis after hepatectomy. World J Gastroenterol 20:126–132
- Hardy KM, Kirschmann DA, Seftor EA, Margaryan NV, Postovit LM, Strizzi L, Hendrix MJ (2010) Regulation of the embryonic morphogen Nodal by Notch4 facilitates manifestation of the aggressive melanoma phenotype. Cancer Res 70:10340–10350
- He G, Dhar D, Nakagawa H, Font-Burgada J, Ogata H, Jiang Y, Shalapour S, Seki E et al (2013) Identification of liver cancer progenitors whose malignant progression depends on autocrine IL-6 signaling. Cell 155:384–396
- Herrera MB, Fonsato V, Gatti S, Deregibus MC, Sordi A, Cantarella D, Calogero R et al (2010) Human liver stem cell-derived microvesicles accelerate hepatic regeneration in hepatectomized rats. J Cell Mol Med 14:1605–1618
- Hess AR, Seftor EA, Gardner LM, Carles-Kinch K, Schneider GB, Seftor RE, Kinch MS et al (2001) Molecular regulation of tumor cell vasculogenic mimicry by tyrosine phosphorylation: role of epithelial cell kinase (Eck/EphA2). Cancer Res 61:3250–3255
- Hess AR, Margarayan NV, Seftor EA, Hendrix MJ (2007)
  Deciphering the signaling events that promote melanoma tumor cell vasculogenic mimicry and their link to embryonic vasculogenesis: role of the Eph receptors.

  Dev Dyn 236:3283–3296
- Ho JW, Pang RW, Lau C, Sun CK, Yu WC, Fan ST, Poon RT (2006) Significance of circulating endothelial progenitor cells in hepatocellular carcinoma. Hepatology 44:836–843
- Ho DW, Yang ZF, Yi K, Lam CT, Ng MN, Yu WC, Lau J, Wan T, Wang X, Yan Z et al (2012) Gene expression profiling of liver cancer stem cells by RNA-sequencing. PLoS One 7, e37159
- Hou Y, Zou Q, Ge R, Shen F, Wang Y (2012) The critical role of CD133(+)CD44(+/high) tumor cells in hematogenous metastasis of liver cancers. Cell Res 22:259–272
- Hugo HJ, Kokkinos MI, Blick T, Ackland ML, Thompson EW, Newgreen DF (2011) Defining the E-cadherin repressor interactome in epithelial-mesenchymal

- transition: the PMC42 model as a case study. Cells Tissues Organs 193:23-40
- Idrissi ME, Hervieu V, Merle P, Mortreux F, Wattel E (2013) Cause-specific telomere factors deregulation in hepatocellular carcinoma. J Exp Clin Cancer Res 32:64
- Imrich S, Hachmeister M, Gires O (2012) EpCAM and its potential role in tumor-initiating cells. Cell Adh Migr 6:30–38
- Ishikawa T, Banas A, Hagiwara K, Iwaguro H, Ochiya T (2010) Stem cells for hepatic regeneration: the role of adipose tissue derived mesenchymal stem cells. Curr Stem Cell Res Ther 5:182–189
- Iwahashi S, Utsunomiya T, Shimada M, Saito Y, Morine Y, Imura S, Ikemoto T, Mori H et al (2013) High expression of cancer stem cell markers in cholangiocellular carcinoma. Surg Today 43:654–660
- Ji J, Yamashita T, Budhu A, Forgues M, Jia HL, Li C, Deng C, Wauthier E, Reid LM et al (2009) Identification of microRNA-181 by genome-wide screening as a critical player in EpCAM-positive hepatic cancer stem cells. Hepatology 50:472–480
- Ji J, Wang XW (2012) Clinical implications of cancer stem cell biology in hepatocellular carcinoma. Semin Oncol 39:461–472
- Jia Y, Liu H, Zhuang Q, Xu S, Yang Z, Li J, Lou J, Zhang W (2012) Tumorigenicity of cancer stem-like cells derived from hepatocarcinoma is regulated by microRNA-145. Oncol Rep 27:1865–1872
- Jia SQ, Ren JJ, Dong PD, Meng XK (2013) Probing the hepatic progenitor cell in human hepatocellular carcinoma. Gastroenterol Res Pract. doi:10.1155/2013/ 145253
- Jing Y, Han Z, Liu Y, Sun K, Zhang S, Jiang G, Li R, Gao L, Zhao X, Wu D, Cai X et al (2012) Mesenchymal stem cells in inflammation microenvironment accelerates hepatocellular carcinoma metastasis by inducing epithelial-mesenchymal transition. PLoS One 7, e43372
- Kamiya A, Gonzalez FJ, Nakauchi H (2006) Identification and differentiation of hepatic stem cells during liver development. Front Biosci 11:1302–1310
- Kamiya A, Inagaki Y (2015) Stem and progenitor cell systems in liver development and regeneration. Hepatol Res 45(1):29–37. doi:10.1111/hepr.12349
- Kamohara Y, Haraguchi N, Mimori K, Tanaka F, Inoue H, Mori M, Kanematsu T (2008) The search for cancer stem cells in hepatocellular carcinoma. Surgery 144:119–124
- Kappel D, Butter F, Benda C, Scheibe M, Draskovic I, Stevense M, Novo CL, Basquin C et al (2013) HOT1 is a mammalian direct telomere repeat-binding protein contributing to telomerase recruitment. EMBO J 32:1681–1701
- Katoonizadeh A, Poustchi H, Malekzadeh R (2014) Hepatic progenitor cells in liver regeneration: current advances and clinical perspectives. Liver Int 34(10): 1464–1472. doi:10.1111/liv.12573
- Kim H, Choi GH, Na DC, Ahn EY, Kim GI, Lee JE, Cho JY, Yoo JE, Choi JS et al (2011) Human hepatocellular

- carcinoms with "stemness"-related marker expression: keratin 19 expression and a poor prognosis. Hepatology 54:1707–1717
- Kim SK, Nasu A, Komori J, Shimizu T, Matsumoto Y, Minaki Y, Kohno K, Shimizu K et al (2014) A model of liver carcinogenesis originating from hepatic progenitor cells with accumulation of genetic alterations. Int J Cancer 134:1067–1076
- Kirschmann DA, Seftor EA, Hardy KM, Seftor RE, Hendrix MJ (2012) Molecular pathways: vasculogenic mimicry in tumor cells: diagnostic and therapeutic implications. Clin Cancer Res 18:2726–2732
- Kitisin K, Pishvaian MJ, Johnson LB, Mishra L (2007) Liver stem cells and molecular signaling pathways in hepatocellular carcinoma. Gastrointest Cancer Res 1(suppl 2):S13–S21
- Klenova et al. (2002) Pub Med PMID 12191639
- Kohga K, Tasumi T, Takehara T, Tsunematsu H, Shimizu S, Yamamoto M, Sasakawa A et al (2010) Expression of CD133 confers malignant potential by regulating metalloproteinases in human hepatocellular carcinoma. J Hepatol 52:872–879
- Kuhlmann WD, Peschke P (2006) Hepatic progenitor cells, stem cells, and AFP expression in models of liver injury. Int J Exp Pathol 87:343–359
- Kupcova SH (2013) Proteomic techniques for characterization of mesenchymal stem cell secretome. Biochimie 95:2196–2211
- Lai CY, Schwartz BE, Hsu MY (2012) CD133+ melanoma subpopulations contribute to perivascular niche morphogenesis and tumorigenicity through vasculogenic mimicry. Cancer Res 72:5111–5118
- Lan X, Wu YZ, Wang Y, Wu FR, Zang CB, Tang C, Cao S, Li SL (2013) CD133 silencing inhibits stemness properties and enhances chemoradiosensitivity in CD133positive liver cancer stem cells. Int J Mol Med 31:315–324
- Lau CK, Yang ZF, Fan ST (2011) Role of stem cells in normal liver and cancer. Anticancer Agents Med Chem 11:522–528
- Lee TK, Poon RT, Yuen AP, Ling MT, Kwok WK, Wang XH, Wong YC, Guan XY et al (2006) Twist overexpression correlates with hepatocellular carcinoma metastasis through induction of epithelial-mesenchymal transition. Clin Cancer Res 12:5369–5376
- Lee TK, Castilho A, Ma S, Ng IO (2009) Liver cancer stem cells: implications for a new therapeutic target. Liver Int 29:955–965
- Lee TH, D'Asti E, Magnus N, Al-Nedawi K, Meehan B, Rak J (2011) Microvesicles as mediators of intracellular communication in cancer-the emerging science of cellular 'debris'. Semin Immunopathol 33:455–467
- Lee JI, Lee JW, Kim JM, Kim JK, Chung HJ, Kim YS (2012a) Prognosis of hepatocellular carcinoma expression cytokeratin 19: comparison with other liver cancers. World J Gastroenterol 18:4751–4757
- Lee Y, El Andaloussi S, Wood MJ (2012b) Exosomes and microvesicles: extracellular vesicles for genetic

- information transfer and gene therapy. Hum Mol Genet 21:R125–R134
- Leelawat K, Tongtawee T, Narong S, Subwongcharoen S, Treepongkaruna SA (2011) Strong expression of CD133 is associated with increased cholangiocarcinoma. World J Gastroenterol 17:1192–1198
- Lehwald N, Duhme C, Wildner M, Kuhn S, Fürst G, Forbes SJ, Jonas S, Robson SC et al (2014) HGF and SDF-1-mediated mobilization of CD133+ BMSC for hepatic regeneration following extensive liver resection. Liver Int 34:89–101
- Li Z (2013) CD133: a stem cell biomarker and beyond. Exp Hematol Oncol 2:17
- Li GC, Ye QH, Dong QZ, Ren N, Jia HL, Qin LX (2013) Mesenchymal stem cells seldomly fuse with hepatocellular carcinoma cells and are mainly distributed in the tumor stroma in mouse models. Oncol Rep 29:713–719
- Lidprapamomgkol K, Chiablaem K, Sila-Asna M, Surarit R, Bunyaratvej A, Svasti J (2012) Exploring stemness gene expression and vasculogenic mimicry capacity in well- and poorly-differentiated hepatocellular carcinoma cell lines. Biochem Biophys Res Commun 422:429–435
- Lin WM, Karsten U, Goletz S, Cheng RC, Cao Y (2011) Expression of CD176 (Thomsen-Friedenreich antigen) on lung, breast and liver cancer-initiating cells. Int J Exp Pathol 92:97–105
- Lingala S, Cui YY, Chen X, Ruebner BH, Qian XF, Zern MA, Wu J (2010) Immunohistochemical staining of cancer stem cell markers in hepatocellular carcinoma. Exp Mol Pathol 89:27–35
- Lirdprapamongkol K, Chiablaem K, Sila-Asna M, Surarit R, Bunyaratvej A, Svasti J (2012) Exploring stemness gene expression and vasculogenic mimicry capacity in well-and poorly-differentiated hepatocellular carcinoma cell lines. Biochem Biophys Res Commun 422:429–435
- Lissitzky JC, Parriaux D, Ristorcelli E, Vérine A, Lombardo D, Verrando P (2009) Cyclic AMP signaling as a mediator of vasculogenic mimicry in aggressive human melanoma cells in vitro. Cancer Res 69:802–809
- Liu D, O'Connor MS, Qin J, Songyang Z (2004) Telosome, a mammalian telomere-associated complex formed by multiple telomeric proteins. J Biol Chem 279:51338–51342
- Liu LL, Fu D, Ma Y, Shen XZ (2011a) The power and the promise of liver cancer stem cell markers. Stem Cells Dev 20:2023–2030
- Liu WB, Xu GL, Jia WD, Li JS, Ma JL, Chen K, Wang ZH, Ge YS, Ren WH et al (2011b) Prognostic significance and mechanisms of patterened matrix vasculogenic mimicry in hepatocellular carcinoma. Med Oncol 28 (suppl 1):S228–S238
- Liu W, Xu G, Ma J, Jia W, Li J, Chen K, Wang W, Hao C, Wang Y, Wang X (2011c) Osteopontin as a key mediator for vasculogenic mimicry in hepatocellular carcinoma. Tohoku J Exp Med 224:29–39

- Liu J, Huang J, Yao WY, Ben QW, Chen DF, He XY, Li L, Yuan YZ (2012) The origins of vascularization in tumors. Front Biosci 17:2559–2565
- Liu S, Li N, Yu X, Xiao X, Cheng K, Hu J, Wang J, Zhang D, Cheng S, Liu S (2013a) Expression of intercellular adhesion molecule 1 by hepatocellular carcinoma stem cells and circulating tumor cells. Gastroenterology 144:1031–1041
- Liu J, Ruan B, You N, Huang Q, Liu W, Dang Z, Xu W, Zhou T, Ji R, Cao Y, Li X et al (2013b) Downregulation of miR-200a induces EMT phenotypes and CSC-like signatures through targeting the b-catenin pathway in hepatic oval cells. PLoS One 8, e79409
- Liu AY, Cai Y, Mao Y, Lin Y, Zheng H, Wu T, Huang Y, Fang X, Lin S, Feng Q, Huang Z et al (2014) Twist2 promotes self-renewal of liver cancer stem-like cells by regulating CD24. Carcinogenesis 35:537–545
- Low KC, Tergaonkar V (2013) Telomerase: central regulator of all of the hallmarks of cancer. Trends Biochem Sci 38:426–434
- Loukinov et al. (2002) Pub Med PMID 12011441
- Ma S, Chan KW, Hu L, Lee TK, Wo JY, Ng IO, Zheng BJ, Guan XY (2007) Identification and characterization of tumorigenic liver cancer stem/progenitor cells. Gastroenterology 132:2542–2556
- Ma S, Lee TK, Zheng BJ, Chan KW, Guan XY (2008) CD133+ HCC cancer stem cells confer chemoresistance by preferential expression of the Akt/PKB survival pathway. Oncogene 27:1749–1758
- Ma S, Tang KH, Chan YP, Lee TK, Kwan PS, Castilho A, Ng I, Man K, Wong N et al (2010) miR-130b promotes CD133(+) liver tumor-initiating cell growth and selfrenewal via tumor protein 53-induced nuclear protein 1. Cell Stem Cell 7:694–707
- Ma JL, Han SX, Zhu Q, Zhao J, Zhang D, Wang L, Lv Y (2011) Role of twist in vasculogenic mimicry formation in hypoxic hepatocellular carcinoma cells in vitro. Biochem Biophys Res Commun 408:686–691
- Ma S (2013) Biology and clinical implications of CD133(+) liver cancer stem cells. Exp Cell Res 319:126–132
- Ma YC, Yang JY, Yan LN (2013) Relevant markers of cancer stem cells indicate a poor prognosis in hepatocellular carcinoma patients: a meta-analysis. Eur J Gastroenterol Hepatol 25:1007–1016
- Machida K, Feldman DE, Tsukamoto H (2015) TLR4dependent tumor-initiating stem cell-like cells (TICs) in alcohol-associated hepatocellular carcinogenesis. Adv Exp Med Biol 815:131–144
- Majumdar A, Curley SA, Wu X, Brown P, Hwang JP, Shetty K, Yao ZX, He AR et al (2012) Hepatic stem cells and transforming growth factor b in hepatocellular carcinoma. Nat Rev Gastroenterol Hepatol 9:530–538
- Mak AB, Blakely KM, Williams RA, Penttilä PA, Shukalyuk AI, Osman KT, Kasimer D et al (2011) CD133 protein N-glycosylation processing contributes to cell surface recognition of the primitive cell marker AC133 epitope. J Biol Chem 286:41046–41056

- Maniotis AJ, Folberg R, Hess A, Seftor EA, Gardner LM, Pe'er J, Trent JM, Meltzer PS et al (1999) Vascular channel formation by human melanoma cells in vivo and in vitro: vasculogenic mimicry. Am J Pathol 155:739–752
- Mansuroglu T, Baumhoer D, Dudas J, Haller F, Cameron S, Lorf T, Füzesi L, Ramadori G (2009) Expression of stem cell factor receptor c-kit in human nontumoral and tumoral hepatic cells. Eur J Gastroenterol Hepatol 21:1206–1211
- Marquardt JU, Thorgeirsson SS (2010) Stem cells in hepatocarcinogenesis: evidence from genomic data. Semin Liver Dis 30:26–34
- Martin-Padura I, Marighetti P, Agliano A, Colombo F, Larzabal L, Redrado M, Bleau AM et al (2012) Residual dormant cancer stem-cell foci are responsible for tumor relapse after antiangiogenic metronomic therapy in hepatocellular carcinoma xenografts. Lab Invest 92:952–966
- Matsuo N, Shiraha H, Fujikawa T, Takaoka N, Ueda N, Tanaka S, Nishina S, Nakanishi Y et al (2009) Twist expression promotes migration and invasion in hepatocellular carcinoma. BMC Cancer 9:240
- McAllister JC, Zhan Q, Weishaupt C, Hsu MY, Murphy GF (2010) The embryonic morphogen, Nodal, is associated with channel-like structures in human malignant melanoma xenografts. J Cutan Pathol 37(Suppl1): 19–25
- Meacham C.E. and Morrison S.J. (2013). Tumour heterogeneity and cancer cell plasticity. Nature 501, 328–337
- Melero-Martin JM, Dudley AC (2011) Concise review: vascular stem cells and tumor angiogenesis. Stem Cells 29:163–168
- Mishra L, Banker T, Murray J, Byers S, Thenappan A, He AR, Shetty K, Johnson L et al (2009) Liver stem cells and hepatocellular carcinoma. Hepatology 49:318–329
- Mizrak D, Brittan M, Allison MR (2008) CD133: molecule of the moment. J Pathol 214:3–9
- Mokarizadeh A, Rezvanfar MA, Dorostkar A, Abdollahi M (2013) Mesenchymal stem cell derived microvesicles: trophic shuttles for enhancement of sperm quality parameters. Reprod Toxicol 42C:78–84
- Moriyama T, Ohuchida K, Mizumoto K, Cui L, Ikenaga N, Sato N, Tanaka M (2010) Enhanced cell migration and invasion of CD133+ pancreatic cancer cells cocultured with pancreatic stromal cells. Cancer 116:3357–3368
- Muralidharan-Chari V, Clancy JW, Sedgwik A, D'Souza-Schorey C (2010) Microvesicles: mediators of extracellular communication during cancer progression. J Cell Sci 123:1603–1611
- Muramatsu S, Tanaka S, Mogushi K, Adikrisna R, Aihara A, Ban D, Ochiai T, Irie T et al (2013) Visualization of stem cell features in human hepatocellular carcinoma reveals in vivo significance of tumor-host interaction and clinical course. Hepatology 58:218–228
- Na DC, Lee JE, Yoo JE, Oh BK, Choi GH, Park YN (2011) Invasion and EMT-associated genes are up-regulated in

- B viral hepatocellular carcinoma with high expression of CD133-human and cell culture study. Exp Mol Pathol 90:66–73
- Nakagawa H, Hikiba Y, Hirata Y, Font-Burgada J, Sakamoto K, Hayakawa Y, Taniguchi K et al (2014) Loss of liver E-cadherin induces sclerosing cholangitis and promotes carcinogenesis. Proc Natl Acad Sci U S A 111:1090–1095
- Nakayama M, Ogasawara S, Akiba J, Ueda K, Koura K, Todoroki K, Kinoshita H, Yano H (2014) Side population cell fractions from hepatocellular carcinoma cell lines increased with tumor dedifferentiation, but lack characteristic features of cancer stem cells. J Gastroenterol Hepatol 29:1092–1101
- Navarro-Alvarez N, Soto-Gutierrez A, Kobayashi N (2010) Hepatic stem cells and liver development. Methods Mol Biol 640:181–236
- Nejak-Bowen KN, Monga SP (2011) Beta-catenin signaling, liver regeneration and hepatocellular cancer: sorting the good from the bad. Semin Cancer Biol 21:44–58
- Nishina H (2012) hDlk-1: a cell surface marker common to normal hepatic stem/progenitor cells and carcinomas. J Biochem 152:121–123
- Oh SH, Witek RP, Bae SH, Zheng D, Jung Y, Piscaglia AC, Petersen BE (2007) Bone marrow-derived hepatic oval cells differentiate into hepatocytes in 2-acetylfluorene/partial hepatectomy-induced liver regeneration. Gastroenterology 132:1077–1087
- Ohnishi S, Maehara O, Nakagawa K, Kameya A, Otaki K, Fujita H, Higashi R, Takagi K et al (2013) Hypoxiainducible factors activate CD133 promoter through ETS family transcription factors. PLoS One 8, e66255
- Oikawa T, Kamiya A, Zeniya M, Chikada H, Hyuck AD, Yamazaki Y, Wauthier E, Tajiri H et al (2013) Sal-like protein 4 (SALL4), a stem cell biomarker in liver cancers. Hepatology 57:1469–1483
- Oishi N, Yamashita T, Kaneko S (2014) Molecular biology of liver cancer stem cells. Liver Cancer 3:71–84
- Oliva J, French BA, Qing X, French SW (2010) The identification of stem cells in human liver diseases and hepatocellular carcinoma. Exp Mol Pathol 88:331–340
- Palm W, de Lange T (2008) How shelterin protects mammalian telomeres. Annu Rev Genet 42:301–334
- Pang RW, Poon RT (2012) Cancer stem cell as a potential therapeutic target in hepatocellular carcinoma. Curr Cancer Drug Targets 12:1081–1094
- Pap E (2011) The role of microvesicles in malignancies. Adv Exp Med Biol 714:183–199
- Papp V, Rokusz A, Dezsö K, Bugyik E, Szabo V, Pavai Z, Paku S, Nagy P (2014) Expansion of hepatic stem cell compartment boosts liver regeneration. Stem Cells Dev 23:56–65
- Park JR, Kim RJ, Lee YK, Kim SR, Roh KJ, Oh SH, Kong G, Kang KS, Nam JS (2011) Dysadherin can enhance tumorigenesis by conferring properties of stem-like cells to hepatocellular carcinoma cells. J Hepatol 54:122–131

- Parveen N, Aleem AK, Habeeb MA, Habibullah CM (2011) An update on hepatic stem cells: bench to bedside. Curr Pharm Biotechnol 12:226–230
- Paulis YW, Soetekouw PM, Verheul HM, Tjan-heijnen VC, Griffoen AW (2010) Signalling pathways in vasculogenic mimicry. Biochim Biophys Acta 1806:18–28
- Petersen BE, Bowen WC, Patrene KD, Mars WM, Sullivan AK, Murase N, Boggs SS et al (1999) Bone marrow as a potential source of hepatic oval cells. Science 284:1168–1170
- Piao LS, Hur W, Kim TK, Hong SW, Kim SW, Choi JE, Sung PS, Song MJ, Lee BC et al (2012) CD133(+) liver cancer stem cells modulate radioresistance in human hepatocellular carcinoma. Cancer Lett 315:129–137
- Ping YF, Bian XW (2011) Concise review: contribution of cancer stem cells to neovascularization. Stem Cells 29:888–894
- Qi W, Liang W, Jiang H, Miuyee WM (2013) The function of miRNA in hepatic cancer stem cell. Biomed Res Int 2013:358902
- Quesenberry PJ, Aliotta JM (2010) Cellular phenotype switching and microvesicles. Adv Drug Deliv Rev 62:1141–1148
- Quesenberry PJ, Dooner MS, Goldberg LR, Aliotta JM, Pereira M, Amaral A, Del Tatto MM et al (2012) A new stem cell biology: the continuum and microvesicles. Trans Am Clin Climatol Assoc 123:152–166
- Rappa G, Mercapide J, Anzanello F, Pope RM, Lorico A (2013) Biochemical and biological characterization of exosomes containing prominin-1/CD133. Mol Cancer 12:62
- Reagan MR, Kaplan DL (2011) Concise review: mesenchymal stem cell tumor-homing: detection methods in disease model systems. Stem Cells 29:920–927
- Rodrigo-Torres D, Affo S, Coll M, Morales-Ibanez O, Millan C, Blaya D, Alvarez-Guaita A et al (2014) The biliary epithelium gives rise to liver progenitor cells. Hepatology 60(4):1367–1377. doi:10.1002/hep.27078
- Rodriguez RM, Fernandez AF, Fraga MF (2013) Role of sirtuins in stem cell differentiation. Genes Cancer 4:105–111
- Rosa-Garrido M, Ceballos L, Alonso-Lecue P, Abraira C, Delgado MD, Gandarillas A (2012) A cell cycle role for the epigenetic factor CTCF-L/BORIS. PLoS One 7, e39371
- Rountree CB, Ding W, Dang H, Vankirk C, Crooks GM (2011) Isolation of CD133+ liver stem cells for clonal expansion. J Vis Exp 56:3183
- Rountree CB, Mishra L, Willenbring H (2012) Stem cells in liver diseases and cancer: recent advances on the path to new therapies. Hepatology 55:298–306
- Saeed H, Iqtedar M (2013) Stem cell function and maintenance – ends that matter: role of telomeres and telomerase. J Biosc 38:641–649
- Sainz B, Heeschen C (2013) Standing out from the crowd: cancer stem cells in hepatocellular carcinoma. Cancer Cell 23:431–433, Cancer Lett 275:185–193
- Sangan CB, Tosh D (2010) Hepatic progenitor cells. Cell Tissue Res 342:131–137

- Sasaki A, Kamiyama T, Yokoo H, Nakanishi K, Kubota K, Haga H, Matsushita M, Ozaki M et al (2010) Cytoplasmic expression of CD133 is an important risk factor for overall survival in hepatocellular carcinoma. Oncol Rep 24:537–546
- Seftor RE, Hess AR, Seftor EA, Kirschmann DA, Hardy KM, Margaryan NV, Hendrix MJ (2012) Tumor cell vasculogenic mimicry: from controversy to therapeutic promise. Am J Pathol 181:1115–1125
- Sell S (1993) The role of determined stem-cells in the cellular lineage of hepatocellular carcinoma. Int J Dev Biol 37:189–201
- Sell S (2008) Alpha-fetoprotein, stem cells and cancer: how study of the production of alpha-fetoprotein during chemical hepatocarcinogenesis led to reaffirmation of the stem cell theory of cancer. Tumour Biol 29:161–180
- Sell S, Leffert HL (2008) Liver cancer stem cells. J Clin Oncol 26:2800–2805
- Shan J, Shen J, Liu L, Xia F, Xu C, Duan G, Xu Y, Ma Q, Yang Z, Zhang Q, Ma L et al (2012) Nanog regulates self-renewal of cancer stem cells through the insulinlike growth factor pathway in human hepatocellular carcinoma. Hepatology 56:1004–1014
- Sharma AD, Cantz T, Manns MP, Ott M (2006) The role of stem cells in physiology, pathophysiology, and therapy of the liver. Stem Cell Rev 2:51–58
- Shen Y, Cao D (2012) Hepatocellular carcinoma stem cells: origins and roles in hepatocarcinogenesis and disease progression. Front Biosc (Elide Ed) 4:1157–1169
- Shi C, Tian R, Wang M, Wang X, Jiang J, Zhang Z, Li X, He Z, Gong W, Qin R (2010) CD44+ CD133+ population exhibits cancer stem cell-like characteristics in human gallbladder carcinoma. Cancer Biol Ther 10:1182–1190
- Shih YT, Wang MC, Zhou J, Peng HH, Lee DY, Chiu JJ (2014) Endothelial progenitors promote hepatocarcinoma intrahepatic metastasis through monocyte chemotactic protein-1 induction of microRNA-21. Gut 64(7):1132–1147. doi:10.1136/gutjnl-2013-306302
- Shin D, Monga SP (2013) Cellular and molecular basis of liver development. Comp Physiol 3:799–815
- Shupe T, Petersen BE (2005) Evidence regarding a stem cell origin of hepatocellular carcinoma. Stem Cell Rev 1:261–264
- Song W, Li H, Tao K, Li R, Song Z, Zhao Q, Zhang F, Dou K (2008) Expression and clinical significance of the stem cell marker CD133 in hepatocellular carcinoma. Int J Clin Pract 62:1212–1218
- Song YJ, Zhang SS, Guo XL, Sun K, Han ZP, Li R, Zhao QD, Deng WJ, Xie XQ et al (2013) Autophagy contributes to the survival of CD133+ liver cancer stem cells in the hypoxic and nutrient-deprived tumor microenvironment. Cancer Lett 339:70–81
- Stock P, Brückner S, Winkler S, Dollinger MM, Christ B (2014) Human bone marrow mesenchymal stem cell-derived hepatocytes improve the mouse liver after acute acetaminophen intoxication by preventing progress of injury. Int J Mol Sci 15:7004–7028

- Strain AJ, Crosby HA, Nijar S, Kelly DA, Hubscher SG (2003) Human liver-derived stem cells. Semin Liver Dis 23:373–384
- Su F, Li B, Wang J, Xu X, Ren R, Li L, Gao F, Liu X (2009) Molecular regulation of vasculogenic mimicry in human uveal melanoma cells: role of helix-loop-helix ld2 (inhibitor of DANN binding 2). Graefes Arch Clin Exp Ophthalmol 247:411–419
- Suetsugu A, Nagaki M, Aoki H, Motohashi T, Kunisada T, Moriwaki H (2006) Characterization of CD133+ hepatocellular carcinoma cells as cancer stem/progenitor cells. Biochem Biophys Res Commun 351:820–824
- Suetsugu A, Osawa Y, Nagaki M, Moriwaki H, Saji S, Bouvet M, Hoffman RM (2010) Simultaneous colorcoded imaging to distinguish cancer "stem-like" and non-stem cells in the same tumor. J Cell Biochem 111:1035–1041
- Sukowati CH, Tiribelli C (2013) The biological implication of cancer stem cells in hepatocellular carcinoma: a possible target for future therapy. Expert Rev Gastroenterol Hepatol 7:749–757
- Sun B, Zhang S, Zhang D, Du J, Guo H, Zhao X, Zhang W, Hao X (2006) Vasculogenic mimicry is associated with high tumor grade, invasion and metastasis, and short survival in patients with hepatocellular carcinoma. Oncol Rep 16:693–698
- Sun B, Zhang D, Zhang S, Zhang W, Guo H, Zhao X (2007) Hypoxia influences vasculogenic mimicry channel formation and tumor invasion-related expression in melanoma. Cancer Lett 249:188–197
- Sun T, Zhao N, Zhao XL, Gu Q, Zhang SW, Che N, Wang XH, Du J, Liu YX, Sun BC (2010) Expression and functional significance of Twist1 in hepatocellular carcinoma: its role in vasculogenic mimicry. Hepatology 51:545–556
- Sun T, Sun BC, Zhao XL, Zhao N, Dong XY, Che N, Yao Z, Ma YM, Gu Q, Zong WK et al (2011) Promotion of tumor cell metastasis and vasculogenic mimicry by way of transcription coactivation by Bcl-2 and Twist1: a study of hepatocellular carcinoma. Hepatology 54:1690–1706
- Sun XT, Yuan XW, Zhu HT, Deng ZM, Yu DC, Zhou X, Ding YT (2012) Endothelial precursor cells promote angiogenesis in hepatocellular carcinoma. World J Gastroenterol 18:4925–4933
- Sun C, Sun L, Jiang K, Gao DM, Kang XN, Wang C, Zhang S, Huang S, Qin X, Li Y et al (2013a) NANOG promotes liver cancer cell invasion by inducing epithelial-mesenchymal transition through NODAL/SMAD3 signaling pathway. Int J Biochem Cell Biol 45:1099–1108
- Sun D, Sun B, Liu T, Zhao X, Che N, Gu Q, Dong X, Yao Z, Li R, Li J, Chi J, Sun R (2013b) Slug promoted vasculogenic mimicry in hepatocellular carcinoma. J Cell Mol Med 17:1038–1047
- Tang KH, Ma S, Lee TK, Chan YP, Kwan PS, Tong CM, Ng IO, Man K, To KF et al (2012) CD133(+) liver tumor-initiating cells promote tumor angiogenesis,

- growth, and self-renewal through neurotensin/interleukin-8/CXCL1 signaling. Hepatology 55:807–820
- Tang J, Tao ZH, Wen D, Wan JL, Liu DL, Zhang S, Cui JF, Sun HC, Wang L, Zhou J et al (2014) miR-612 suppresses the stemness of liver cancer via Wnt/b-catenin signaling. Biochem Biophys Res Commun 447:210–215
- Terris B, Cavard C, Perret C (2010) EpCAM, a new marker for cancer stem cells in hepatocellular carcinoma. J Hepatol 52:280–281
- Teufel A, Galle PR (2010) Collecting evidence for a stem cell hypothesis in HCC. Gut 59:870–871
- Tiffen JC, Bailey CG, Marshall AD, Metierre C, Feng Y, Wang Q, Watson SL, Holst J et al (2013) The cancertestis antigen BORIS phenocopies the tumor suppressor CTCF in normal and neoplastic cells. Int J Cancer 133:1603–1613
- Tomuleasa C, Soritau O, Rus-Ciuca D, Pop T, Todea D, Mosteanu O, Pintea B, Foris V et al (2010) Isolation and characterization of hepatic cancer cells with stem-like properties from hepatocellular carcinoma. J Gastrointestin Liver Dis 19:61–67
- Tong CM, Ma S, Guan XY (2011) Biology of hepatic cancer stem cells. J Gastroenterol Hepatol 26:1229–1237
- Toshima T, Yoshizumi T, Uchiyama H, Ikegami T, Soejima Y, Ikeda T, Kawanaka H et al (2013) Effect of Cd133-positive stem cells in repeated recurrence of hepatocellular carcinoma after liver transplantation: a case report. Fukuoka Igaku Zasshi 104:383–388
- Tsai ST, Tsou CC, Mao WY, Chang WC, Han HY, Hsu WL, Li CL, Shen CN et al (2012) Label-free quantitative proteomics of CD133-positive liver cancer stem cells. Proteome Sci 10:69
- Tsuchiya A, Kamimura H, Tamura Y, Takamura M, Yamagiwa S, Suda T, Nomoto M et al (2011) Hepatocellular carcinoma with progenitor cell features distinguishable by the hepatic stem/progenitor cell marker NCAM. Cancer Lett 309:95–103
- Valyi-Nagy K, Kormos B, Ali M, Shukla D, Valyi-Nagy T (2012) Stem cell marker CD271 is expressed by vasculogenic mimicry-forming uveal melanoma cells in three-dimensional cultures. Mol Vis 18:588–592
- Vartanian A, Gatsina G, Grigorieva I, Solomko E, Dombrovsky V, Baryshnikov A, Stepanova E (2013) The involvement of Notch signaling in melanoma vasculogenic mimicry. Clin Exp Med 13:201–209
- Vessey CJ, de la Hall PM (2001) Hepatic stem cells: a review. Pathology 33:130–141
- Vestentoft PS (2013) Development and molecular composition of the hepatic progenitor cell niche. Dan Med J 60:B4640
- Wang B, Jacob ST (2011) Role of cancer stem cells in hepatocarcino-genesis. Genome Med 3:11
- Wang H, Chen L (2013) Tumor microenvironment and hepatocellular carcinoma metastasis. J Gastroenterol Hepatol 28(suppl 1):43–48
- Wang L, Guo ZY, Zhang R, Xin B, Chen R, Zhao J, Wang T, Wen WH, Jia LT, Yao LB et al (2013)

References 25

Pseudogene OCT4-pg4 functions as a natural micro RNA sponge to regulate OCT4 expression by competing for miR-145 in hepatocellular carcinoma. Carcinogenesis 34:773–781

- Wang F, Yuan JH, Wang SB, Yang F, Yuan SX, Ye C, Yang N, Zhou WP, Li WL, Li W, Sun SH (2014) Oncofetal long noncoding RNA PVT1 promotes proliferation and stem cell-like property of hepatocellular carcinoma cells by stabilizing NOP2. Hepatology 60:1278–1290
- Wells RG (2014) The portal fibroblast Not just a poor man's stellate cell. Gastroenterology 147(1):41–47. doi:10.1053/j.gastro.2014.05.001
- Woo HG, Wang XW, Budhu A, Kim YH, Kwon SM, Tang ZY, Sun Z, Harris CC et al (2011) Association of TP53 mutations with stem cell-like gene expression and survival of patients with hepatocellular carcinoma. Gastroenterology 140:1063–1070
- Wu XZ, Chen D (2006) Origin of hepatocellular carcinoma: role of stem cells. J Gastroenterol Hepatol 21:1093–1098
- Wu Y, Wu PY (2009) CD133 as a marker for cancer stem cells: progresses and concerns. Stem Cells Dev 18:1127–1134
- Wu KJ, Yang MH (2011) Epithelial-mesenchymal transition and cancer stemness: the Twist1-Bmi1 connection. Biosci Rep 31:449–455
- Wu K, Ding J, Chen C, Sun W, Ning BF, Wen W, Huang L, Han T, Yang W, Wang C et al (2012) Hepatic transforming growth factor beta gives rise to tumorinitiating cells and promotes liver cancer development. Hepatology 56:2255–2267
- Xin H, Liu D, Songyang Z (2008) The telosome/shelterin complex and its functions. Genome Biol 9:232
- Xiong W, Sun LP, Chen XM, Li HY, Huang SA, Jie SH (2013) Comparison of microRNA expression profiles in HCC-derived microvesicles and the parental cells and evaluation of their roles in HCC. J Huazhong Univ Sci Technolog Med Sci 33:346–352
- Xu XL, Xing BC, Han HB, Zhao W, Hu MH, Xu ZL, Li JY, Xie Y, Gu J, Wang Y et al (2010a) The properties of tumor-initiating cells from a hepatocellular carcinoma patient's primary and recurrent tumor. Carcinogenesis 31:167–174
- Xu X, Xing B, Hu M, Xu Z, Xie Y, Dai G, Gu J, Wang Y, Zhang Z (2010b) Recurrent hepatocellular carcinoma cells with stem cell-like properties: possible targets for immunotherapy. Cytotherapy 12:190–200
- Xu Y, Xie Y, Wang X, Chen X, Liu Q, Ying M, Zheng Q (2013) Identification of cancer stem cells from hepatocellular carcinoma cell lines and their related microRNAs. Oncol Rep 30:2056–2062
- Xue X, Zhang Y, Zhi Q, Tu M, Xu Y, Sun J, Wei J, Lu Z, Miao Y, Gao W (2014) MiR200-upregulated vasohibin 2 promotes the malignant transformation of tumors by inducing epithelial-mesenchymal transition in hepatocellular carcinoma. Cell Commun Signal 12:62
- Yamashita T, Ji J, Budhu A, Forgues M, Yang W, Wang HY, Jia H, Ye Q, Qin LX et al (2009) EpCAM-positive

- hepatocellular carcinoma cells are tumor-initiating cells with stem/progenitor cell features. Gastroenterology 136:1012–1024
- Yamashita T, Wang XW (2013) Cancer stem cells in the development of liver cancer. J Clin Invest 123:1911–1918
- Yamashita T, Honda M, Nakamoto Y, Baba M, Nio K, Hara Y, Zeng SS, Hayashi T et al (2013) Discrete nature of EpCAM+ and CD90+ cancer stem cells in human hepatocellular carcinoma. Hepatology 57:1484–1497
- Yamashita T, Kaneko S (2014) Orchestration of hepatocellular carcinoma development by diverse liver cancer stem cells. J Gastroenterol 49:1105–1110
- Yan H, Dong X, Zhong X, Ye J, Zhou Y, Yang X, Shen J, Zhang J (2013) Inhibitions of epithelial to mesenchymal transition and cancer stem cells-like properties are involved in miR-148a-mediated anti-metastasis of hepatocellular carcinoma. Mol Carcinog 53(12): 960–969. doi:10.1002/mc.22064
- Yanai H, Nakamura K, Hijoka S, Kamei A, Ikari T, Ishikawa Y, Shinozaki E, Mizunuma N et al (2010) Dlk-1, a cell surface antigen on fetal hepatic stem/ progenitor cells, is expressed in hepatocellular, colon, pancreas and breast carcinomas at a high frequency. J Biochem 148:85–92
- Yang W, Yan HX, Chen L, Liu Q, He YQ, Yu LX, Zhang SH, Huang DD, Tang L et al (2008) Wnt/beta-catenin signaling contributes to activation of normal and tumorigenic liver progenitor cells. Cancer Res 68:4287–4295
- Yang ZF, Poon RT (2008) Vascular changes in hepatocellular carcinoma. Anat Rec (Hoboken) 291:721–734
- Yang XR, Xu Y, Yu B, Zhou J, Qiu SJ, Shi GM, Zhang BH, Wu WZ, Shi YH, Wu B et al (2010) High expression levels of putative hepatic stem/progenitor cell biomarkers related to tumour angiogenesis and poor prognosis of hepatocellular carcinoma. Gut 59:953–962
- Yang W, Wang C, Lin Y, Liu Q, Yu LX, Tang L, Yan HX, Fu J, Chen Y, Zhang HL et al (2012) OV6+ tumorinitiating cells contribute to tumor progression and invasion in human hepatocellular carcinoma. J Hepatol 57:613–620
- Yao Z, Mishra L (2009) Cancer stem cells and hepatocellular carcinoma. Cancer Biol Ther 8:1691–1698
- Yao XH, Ping YF, Bian XW (2011) Contribution of cancer stem cells to tumor vasculogenic mimicry. Protein Cell 2:266–272
- Ye J, Wu Y, Gilson E (2010) Dynamics of telomeric chromatin at the crossroads of aging and cancer. Essays Biochem 48:147–164
- Yeh CT, Kuo CJ, Lai MW, Chen TC, Lin CY, Yeh TS, Lee WC (2009) CD133-positive hepatocellular carcinoma in an area endemic for hepatitis B virus infection. BMC Cancer 9:324
- Yi SY, Nan KJ (2008) Tumor-initiating stem cells in liver cancer. Cancer Biol Ther 7:325–330
- Yin S, Li J, Hu C, Chen X, Yao M, Yan M, Jiang G, Ge C, Xie H, Wan D, Yang S et al (2007) CD133 positive

- hepatocellular carcinoma cells possess high capacity for tumorigenicity. Int J Cancer 120:1444–1450
- Yin C, Evason KJ, Asahina K, Stainier DY (2013) Hepatic stellate cells in liver development, regeneration, and cancer. J Clin Invest 123:1902–1910
- Yong KJ, Gao C, Lim JS, Yan B, Yang H, Dimitrov T, Kawasaki A, Ong CW, Wong KF et al (2013) Oncofetal gene SALL4 in aggressive hepatocellular carcinoma. N Engl J Med 368:2266–2276
- Yoon SK (2012) The biology of cancer stem cells and its clinical implication in hepatocellular carcinoma. Gut Liver 6:29–40
- Yoshikawa S, Zen Y, Fukii T, Sato Y, Ohta T, Aoyagi Y, Nakanuma Y (2009) Characterization of CD133+ parenchymal cells in the liver: histology and culture. World J Gastroenterol 15:4896–4906
- You H, Ding W, Rountree CB (2010) Epigenetic regulation of cancer stem cell marker CD133 by transforming growth factor-beta. Hepatology 51:1635–1644
- Yu DC, Chen J, Sun XT, Zhuang LY, Jiang CP, Ding YT (2010a) Mechanism of endothelial progenitor cell recruitment into neo-vessels in adjacent non-tumor tissues in hepatocellular carcinoma. BMC Cancer 10:435
- Yu DC, Chen J, Ding YT (2010b) Hypoxic and highly angiogenic non-tumor tissues surrounding hepatocellular carcinoma: the 'niche' of endothelial progenitor cells. Int J Mol Sci 11:2901–2909
- Yu XH, Xu LB, Liu C, Zhang R, Wang J (2011) Clinicopathologic characteristics of 20 cases of hepatocellular carcinoma with bile duct tumor thrombi. Dig Dis Sci 56:252–259
- Yu G, Jing Y, Kou X, Ye F, Gao L, Fan Q, Yang Y, Zhao Q, Li R, Wu M, Wei L (2013a) Hepatic stellate cells secreted hepatocyte growth factor contributes to the chemoresistance of hepatocellular carcinoma. PLoS One 8, e73312
- Yu B, He ZY, You P, Han QW, Xiang D, Chen F, Wang MJ, Liu CC, Lin XW, Borjigin U et al (2013b) Reprogramming fibroblasts into bipotential hepatic stem cells by defined factors. Cell Stem Cell 13:328–340
- Yuan A, Farber EL, Rapoport AL, Tejada D, Deniskin R, Akhmeddov NB, Farber DB (2009) Transfer of microRNAs by embryonic stem cell microvesicles. PLoS One 4, e4722
- Yun WJ, Shin E, Lee K, Jung HY, Kim SH, Park YN, Yu E, Jang JJ (2013) Clinicopathologic implication of hepatic progenitor cell marker expression in hepatoblastoma. Pathol Res Pract 209:568–573
- Zenali MJ, Tan D, Li W, Dhingra S, Brown RE (2010) Stemness characteristics of fibrolamellar hepatocellular carcinoma: immunohistochemical analysis with comparisons to comparisons to conventional hepatocellular carcinoma. Ann Clin Lab Sci 40:126–134
- Zhai JM, Yin XY, Hou X, Hao XY, Cai P, Liang LJ, Zhang LJ (2013) Analysis of the genome-wide DNA methylation profile of side population cells in hepatocellular carcinoma. Dig Dis Sci 58:1934–1947
- Zhang A, London R, Schulz FM, Giguère-Simmonds PW, Delriviere L, Chandraratana H et al (2010) Human liver

- progenitor cell lines are readily established from non-tumorous tissue adjacent to hepatocellular carcinoma. Stem Cells Dev 19:1277–1284
- Zhang J, Luo N, Luo Y, Peng Z, Zhang T, Li S (2012) microRNA-150 inhibits human CD133-positive liver cancer stem cells through negative regulation of the transcription factor c-Myb. Int J Oncol 40:747–756
- Zhang C, Xu Y, Zhao J, Fan L, Jiang G, Li R, Ling Y, Wu M, Wie L (2011) Elevated expression of the stem cell marker CD133 associated with Line-1 demethylation in hepatocellular carcinoma. Ann Surg Oncol 18:2373–2380
- Zhao XL, Du J, Zhang SW, Liu XY, Wang X, Sun BC (2006) A study on vasculogenic mimicry in hepatocellular carcinoma (in Chinese). Zhonghua Gan Zang Bing Za Zhi 14:41–44
- Zhao J, Huang JS, Yang AJ, Wang CY, Liu W, Li M (2007)
  Three-dimension cell culture and histology of vasculogenic mimicry in hepatocellular carcinoma (in Chinese). Ai Zheng 26:123–126
- Zhao N, Sun BC, Sun T, Ma YM, Zhao XL, Liu ZY, Dong XY, Che N, Mo J, Gu Q (2012) Hypoxia-induced vasculogenic mimicry formation via VE-cadherin regulation by Bcl-2. Med Oncol 29:3599–3607
- Zheng YW, Taniguchi H (2003) Diversity of hepatic stem cells in the fetal and adult liver. Semin Liver Dis 23:337–348
- Zheng T, Wang J, Jiang H, Liu L (2011) Hippo signaling in oval cells and hepatocercinogenesis. Cancer Lett 302:91–99
- Zheng YW, Nie YZ, Taniguchi H (2013) Cellular reprogramming and hepatocellular carcinoma development. World J Gastroenterol 19:8850–8860
- Zhou L, Yang ZX, Song WJ, Li QJ, Yang F, Wang DS, Zhang N, Dou KF (2013) MicroRNA-21 regulates the migration and invasion of a stem-like population in hepatocellular carcinoma. Int J Oncol 43:661–669
- Zhu Z, Hao X, Yan M, Yao M, Ge C, Gu J, Li J (2010) Cancer stem/progenitor cells are highly enriched in CD133+ CD44+ population in hepatocellular carcinoma. Int J Cancer 126:2067–2078
- Zhu H, Shao Q, Sun X, Deng Z, Yuan X, Yu D, Zhou X, Ding Y (2012a) The mobilization, recruitment and contribution of bone marrow-derived endothelial progenitor cells to the tumor neovascularization occur at an early stage and throughout the entire process of hepatocellular carcinoma growth. Oncol Rep 28:1217–1224
- Zhu H, Shao Q, Sun X, Deng Z, Yuan X, Zhou X, Ding Y (2012b) Bone marrow cells: important role on neovascularization of hepatocellular carcinoma. J Gastroenterol Hepatol 27:1241–1251
- Zimmermann A (2005) The emerging family of hepatoblastoma tumours: from ontogenesis to oncogenesis. Eur J Cancer 41:1503–1514
- Zou GM (2008) Cancer initiating cells or cancer stem cells in the gastrointestinal tract and liver. J Cell Physiol 217:598–604

## 2

# Hepatocellular Carcinoma (Ordinary Hepatocellular Carcinoma)

ICD-O code 8170/3

Contents		Cytologic Features	
Introduction	28	Cell Polarity	45 45
Epidemiology	28	Tumor Cell Inclusions	
-		Bile Accumulation	
Clinical Features  General Features  Paraneoplastic Syndromes	29 29 30	Accumulation of Neutral Fat (Fatty Change, Steatotic Change) and Other Cytoplasmic Changes	16
Complications	30	č	
Imaging Eastures	31	Reticulin Fiber Patterns	47
Imaging Features         Pathology         Introduction	32 32	Loss of Kupffer Cells (Hepatic Macrophages) in Hepatocellular Carcinoma	48 48
Classifications	32 32	Grading of Hepatocellular Carcinoma  Stroma Formation (Desmoplasia) in Hepatocellular Carcinoma	
and Growth Patterns	32 34	Vascular Patterns (Angioarchitecture) of Hepatocellular Carcinoma	50
Macroscopic Pathology of Main Gross Patterns of HCC  Size of Hepatocellular Carcinoma  Nodular HCCs  Massive Hepatocellular Carcinoma  Diffuse Hepatocellular Carcinoma  Pedunculated Hepatocellular Carcinoma	34 34 34 37 37 38	Encapsulation (Pseudocapsule or Capsule Formation) Introduction Pathology Biologic Significance of HCC Pseudocapsules Pathogenesis of Pseudocapsules	
Satellite Lesions	40 40	Differential Diagnosis of Ordinary Hepatocellular Carcinoma	54
Histopathology	41 41 41	References	54
Carcinoma	44		

#### **Abstract**

Hepatocellular carcinoma (HCC) is a malignant neoplasm composed of cells with a hepatocellular differentiation. Most HCCs exhibit a trabecular and/or acinar histologic pattern (ordinary or classical HCC), but there also exist numerous variants with a different morphology and a fibrolamellar variant treated in a separate chapter. HCC is a frequent malignancy and is estimated to be the fifth most common malignancy in males worldwide. Most of the tumors arise in patients with liver cirrhosis caused by various etiologies, hepatitis virus infections, and toxic agents playing a central role. HCC displays specific macroscopic growth patterns. Eggel divided the neoplasms in nodular, massive, and diffuse types, nodular HCC being the most common. This classification has subsequently been extended and refined. HCC frequently invades large and small hepatic vessels and bile ducts, causing distinct clinical disorders. Histologically, hepatocyte-like cells arranged in large plates with an abnormal reticulin pattern and loss of Kupffer cells are the hallmarks of trabecular HCC, whereas pseudoglandular structures are found in other tumors. The grading of HCC has been standardized and is an important prognosticator. Morphological classifications of HCC are currently extended by the use of molecular signatures of these neoplasms.

#### Introduction

Hepatocellular carcinoma (HCC) is a malignant neoplasm composed of cells with a hepatocellular differentiation. The majority of HCC shows trabecular and/or pseudoglandular/acinar patterns that consist of a hepatocyte-like cell lineage (ordinary, "classical," or "standard" HCC). In addition, there exist numerous variants of ordinary HCC and a distinct entity, fibrolamellar hepatocellular carcinoma, treated in separate chapters.

Eggel (1901) was the first to clearly state that primary liver cell cancer is derived from hepatocytes. Apart from the morphologic resemblance between hepatocytes and hepatoid cells in moderately to well-differentiated HCC, a crucial argument of Eggel was his discovery of bile depositions in 6 of the 117 histologically examined tumors that he compiled from the literature, by adding a case of his own. Identification of bile in and between tumor cells still is a hallmark for the diagnosis of HCC.

#### **Epidemiology**

Liver cancer is a frequent malignancy worldwide with an incidence that is particularly high in all low-resource regions of the world, except Northern Africa and Western Asia. It is estimated that 82 % of HCC occur in developing countries (Leong and Leong 2005; Chuang et al. 2009). Estimates from the year 2004 show that HCC is the fifth most common malignancy in males and the eighth in females worldwide (Table 1).

About 564,000 new cases are diagnosed per year, with 398,000 in men and 166,000 in females, illustrating that HCC is two to four times more common in men than in women (reviews: Bosch et al. 2004, 2005). The yearly incidence comprises between 2.5 % and 7 % of patients with liver cirrhosis (Montalto et al. 2002). The highest reported incidence of 100 cases per 100,000 of population annually had been reported from Mozambique, due to high prevalence of HBV infection followed by cirrhosis, aflatoxin

**Table 1** Age-adjusted incidence of HCC per 100,000 inhabitants (Llovet et al. 2003; modified)

	Age-adjusted incidence (males/
Area	females)
Worldwide	14.9/5.5
East Asia	35.4/12.6
Middle Africa	24.2/12.9
Southeast Asia	18.3/5.7
Southern	9.8/3.4
Europe	
Western	5.8/1.6
Europe	
South America	4.8/3.6
North America	4.1/1.6
Northern	2.6/1.3
Europe	

Clinical Features 29

and alcohol toxicity, and possible iron overload. The peak age of onset of HCC is continuously rising worldwide, and there is evidence that the clinical features in the elderly differ from those in younger individuals. In a study of 622 patients with HCC, including 91 patients 70 years old or older, the proportion of females increased, and tumor sizes at diagnosis were smaller in the elderly than in younger patients, whereas clinical stage taking liver function into consideration was similar in the two age groups (Nomura et al. 1994). Whereas an increase in incidence has been noted in certain regions, the incidence of HCC has, e.g., been static over recent decades in the Asia Pacific region (Yuen et al. 2009).

HCC is closely associated with liver cirrhosis (Siegenbeek van Heukelom 1894; Kew and Popper 1984; Gall 1960; Fattovich et al. 2004), the main causes of this chronic liver disease being hepatitis virus infections, alcoholic liver disease, and nonalcoholic fatty liver disease (NAFLD). The development of HCC in a background of cirrhosis varies as a function of underlying disease. Particularly high incidences of HCC in cirrhotic liver are found in HCV infection and hereditary hemochromatosis. HCV infection was found to have the highest HCC incidence in cirrhotic patients, with a 5-year cumulative incidence of 17 % in Western countries and 30 % in Japan. In the cirrhotic stage of hemochromatosis, the 5-year cumulative incidence was 21 %. Lower values are observed in other important causes of cirrhosis, i.e., HBV infection, alcoholic liver disease, and various forms of biliary cirrhosis. In high endemic areas of HBV infection, the 5-year cumulative incidence of HCC is 15 %, while it is 10 % in Western countries. Overall, alcoholic liver cirrhosis has a 5-year cumulative HCC risk of 8 % (review: Fattovich et al. 2004). Male gender and age more than 50 years are risk factors for HCC in cirrhotic patients. A subset of HCC develops in the absence of cirrhosis (Tabarin et al. 1987; Giarelli et al. 1991). In an autopsy study from Japan, non-cirrhotic cases among 618 HCC patients amounted to 10.7 % (Okuda et al. 1989). In Western countries, HCC in non-fibrotic livers was associated with younger age and female sex (Bège et al. 2007).

It seems that the earliest true primary HCC was a case reported in Frerich's series in 1861, followed by a bona fide example of HCC in a cirrhotic liver with invasion of the inferior vena cava, reported in 1876 (Weigert 1876). It is striking to note that in early works on the incidence of HCC in Europe, relatively few cases were found in a clinical setting or in autopsy series in comparison with metastatic liver disease. In his work on liver tumors, Thöle (1913) cites that von Hansemann (a pathologist, assistant of Virchow, who formulated the concept of anaplasia) found 225 metastases and 6 HCC among 258 malignancies observed until 1889 and that Ahlenstiel detected 95 hepatic malignancies in 6,000 autopsies, of which 90 were metastases and 5 primary liver cancer. Similarly, Mau identified, in 1901, 246 hepatic malignancies among 8,587 autopsies performed in Hamburg, Germany, including 242 metastases and 4 primary liver cancers. Riesenfeld (1868) diagnosed liver cancer in 2.3 % of autopsies performed in Berlin between 1864 and 1868. Among 24,400 consecutive autopsies performed at Bellevue Hospital, New York, from 1906 to 1936, 62 cases of primary hepatic carcinoma were identified (Gustafson 1937). Between 1850 and 1950, many publications were written on single cases and small series, and numerous later studies and reviews have then addressed issues of occurrence patterns and epidemiology of HCC.

Selected References Anthony 1973; Sabourin 1881; Okuda et al. 1985; Okuda 1990; Dominguez-Malagon and Gaytan-Graham 2001; El-Serag 2002; Llovet et al. 2003; Bosch et al. 2004, 2005; Srivatanakul et al. 2004; Lee et al. 2009; McGlynn and London 2011.

#### Clinical Features

#### **General Features**

In early HCC, many patients are asymptomatic. In the absence of distinct complications, symptoms and signs in symptomatic patients are nonspecific and in part related to underlying chronic liver disease, in particular hepatic cirrhosis. They include upper abdominal discomfort or pain, effects of hepatomegaly, malaise, anorexia, weight loss, and anemia (Berman 1959). In one European study, malaise (85 %), weight loss (78 %), anorexia (67 %), and hepatomegaly (84 %) were common findings at presentation (Kaczynski et al. 2005). In advanced disease, signs of hepatic failure and jaundice may ensue. In patients with underlying cirrhosis, sequelae of portal hypertension are often the leading clinical signs. Serologically, several markers for HCC and HCC recurrence have been identified, but AFP and its variants remains the most frequently used tumor marker for the diagnosis of HCC (Vogel et al. 1974; Gonzalez and Keeffe 2011; Nakao and Ichikawa 2013). However, a significant proportion of HCC patients do not show elevated serum AFP levels (low-AFP HCC; Carr et al. 2010). Serum AFP is employed for monitoring patients' response to therapy. Specifically, monitoring of Lens culinaris agglutinin-reactive AFP (AFP-L3, the fucosylated variant of AFP) is useful for early detection of recurrent HCC (Okuda 2000; Giannelli and Antonaci 2006). Elevated serum AFP is not only a bystander, but the protein exerts distinct functions. HCC patients with high or very high AFP serum levels show a higher mortality rate, probably related to a tumor growthpromoting activity of AFP (Li et al. 2011). It was earlier reported that AFP has immunosuppressive properties and induces spontaneous T and B lymphocyte responses in HCC patients. These responses are related to distinct immunogenic epitopes of the AFP molecule (review: Bei and Mizejewski 2011). AFP is also capable to activate monocytes and to augment phagocytic capacity of both monocytes and granulocytes (Kong et al. 2012), but abolishes the activation of natural killer cells by inhibiting the function of dendritic cells (Yamamoto et al. 2011).

Assessment of des-gamma-carboxy prothrombin (PIVKA-II), an abnormal prothrombin secreted by HCC, by sensitive assays has an important place in the diagnosis and follow-up of HCC (Takikawa et al. 1992; Bertino

et al. 2012). In one study of 116 HCC, PIVKA-II was detected in serum in 54.3 % of patients and the concentration showed a positive correlation with tumor size (Takikawa et al. 1992). Further serum markers for HCC include glypican-3 and Golgi protein 73 (Giannelli and Antonaci 2006; Tian et al. 2011). Serum carcinoembryonic antigen (CEA) is raised in part of HCC patients (Macnab et al. 1978). For small or early HCC, reliable serum markers have not yet been established (Talwalkar and Gores 2004).

#### **Paraneoplastic Syndromes**

Hepatocellular carcinomas can be associated with a wide array of paraneoplastic syndromes (Table 2).

#### **Complications**

HCC may cause several types of complications. Rupture of HCC is a particularly severe

 Table 2
 Paraneoplastic syndromes associated with hepatocellular carcinoma

Hematologic paraneoplastic syndromes

Erythrocytosis (polyglobulism, erythremia)
Thrombocytosis
Leukemoid reactions (paraneoplastic neutrocytosis)
Peripheral hypereosinophilia
Paraneoplastic dermatological syndromes
(paraneoplastic dermadromes)
Prurigo
Papuloerythroderma
Lichen planus
Lichen myxedematosus
Psoriasiform dermatoses (psoriasis guttata)
Pemphigus-like disorders
Lupus erythematosus-like syndromes
Acanthosis nigricans
Acquired perforating dermatoses
Papuloerythroderma of Ofuji
Acrokeratosis paraneoplastica (Bazex syndrome)
Disseminated porokeratosis
Pityriasis rotunda
Acquired porphyria

(continued)

Imaging Features 31

#### Table 2 (continued)

Neurological paraneoplastic syndromes
Necrotizing leukoencephalopathy
Demyelinating polyradiculoneuropathy
Encephaloradiculopathy
Cerebellar ataxia
Necrotizing myelopathy
Demyelinating polyneuropathy
Peripheral neuropathy
Retinopathy
Autoimmune paraneoplastic syndromes
Dermatomyositis/polymyositis
Autoimmune arthritis/polyarthritis
Myasthenia gravis and other muscle weaknesses
Autoimmune thyroid disease
Behçet's disease
Metabolic disorders
Hypoglycemia
Hypercalcemia
Hypophosphatemia
Paraneoplastic hyperlipidemia and hypercholesterolemia
Hormonal syndromes
Paraneoplastic hyperthyroxinemia
Inappropriate ADH secretion
Coagulation disorders
Acquired von Willebrand disease
Cryofibrinogenemia
Combined paraneoplastic syndromes
Fever of unknown origin (FUO)

complication of this neoplasm and bears a high mortality. Rupture may occur spontaneously or following blunt abdominal trauma (Ong and Taw 1972; Chearanai et al. 1983; Van Landingham et al. (1985); Clarkston et al. (1988); Kanematsu et al. (1992); Chen et al. (1996); Castells et al. (2001); Chedid et al. 2001; Polat et al. 2005; Lai and Lau 2006; Kim et al. 2008). Spontaneous rupture of HCC can also occur in pediatric HCC (Nejmeddine et al. 2010). Localization of tumor to the left lobe and expansion of HCC outside the liver surface are risk factors for tumor rupture (Chen et al. 1995). Spontaneous rupture of HCC with hemoperitoneum is a rare condition in Western countries (Pombo et al. 1991; Chedid et al. 2001), while over 10 % of patients with HCC may experience tumor rupture in certain regions of Africa and Asia (Nagasue and Inokushi

1979). Spontaneous rupture of HCC was found to be the cause of death in only 3 % of patients in one study of 530 HCC patients (Kaczynski et al. 2005). Disruption of the tumor surface and a tear in a parasitic feeding artery are typical causes of HCC rupture (Chen et al. 1995; Kim et al. 2008). Vascular injuries that favor spontaneous rupture include degradation of collagen in the vascular wall (Zhu et al. 2001, 2002) and immune complex-induced damage of elastic lamellae (Zhu et al. 2004, 2006). Risk factors for HCC rupture included underlying disease such as hypertension and cirrhosis, a tumor size exceeding 5 cm, tumor protrusion from the liver surface, vascular thromextrahepatic invasion and (Miyoshi et al. 2011; Zhu et al. 2012). HCC rupture can cause massive hemoperitoneum (Gibbons and Bell 1963). Parts of HCC present with jaundice, which is a negative prognosticator. In a minority of jaundiced patients, hyperbilirubinemia is due to biliary obstruction caused by the tumor, a situation outlined in more detail below. In the majority of patients, however, jaundice is caused by hepatic insufficiency caused by tumor and/or associated liver cirrhosis. Among 530 HCC patients with clinically detectable jaundice, 481 had jaundice due to hepatic insufficiency and 49 patients had obstructive jaundice. Patients with hepatic insufficiency had extremely poor prognosis, and 90 % of them died within 10 weeks of first presentation (Lau et al. 1997).

#### **Imaging Features**

Imaging features of HCC have been described in detail in numerous publications and are not specifically dealt with in this chapter (Okuda 1980a; Vermess et al. 1985; Stevens et al. 1996; Lee et al. 2011). Diagnostic imaging of HCC has recently undergone marked progress, mainly due to the introduction of the ultrasound contrast agent Sonazoid used in contrast-enhanced US (CEUS) and MR techniques employing a liver-specific gadolinium MR contrast agent (Ogawa et al. 2006; Xu et al. 2011; Tanaka et al. 2014; review: Kudo 2011).

#### **Pathology**

#### Introduction

As outlined in the paragraph on classification, HCCs are characterized by several distinct macroscopic patterns. The preparation and documentation of specimens for the identification of gross features has been standardized (Ruby 2000), as has the reporting of tissues removed (Dabbs et al. 2004; Association of Directors of Anatomic and Surgical Pathology 2005). Similarly, the definitions and nomenclature of nodular liver cell lesions has been standardized in a consensus panel (International Working Party 1995). In this paragraph, the morphology of advanced HCC is discussed, while the pathology of early and small HCC (Wanless 2007) is discussed in a separate chapter.

**Selected References** Rindfleisch 1878; Eggel 1901; Wegelin 1905; Rosenberg and Ochsner 1948; Edmondson and Steiner 1954; Edmondson 1955; Berman 1959; Patton and Horn 1964; Foster and Berman 1977; Mori et al. 1980; Anthony 1999; Kai et al. 2012.

Both the macroscopic growth patterns and the histopathologic presentation of HCC vary as a function of geography and genetic background, and between individual patients (Okuda et al. 1984; Okuda 1997). There are certain geographic specificities in regard to the presentation of diffuse and other variants of HCC and the type of underlying liver disease. This issue has been outlined by Okuda (1997) who emphasized that there are significant global variations of clinicopathologic features of HCCs, based on comparative studies of patients from Japan, Los Angeles, and Pretoria in South Africa (Okuda et al. 1984). For example, most patients from Japan show HCCs growing in an expanding fashion, with formation of a fibrous capsule, while capsule formation was very uncommon in South African Blacks. Depending on its gross growth patterns, primary HCC can involve one liver lobe or both liver lobes, whereby bilobar involvement had been found more frequently in HCC not associated with cirrhosis (Ho et al. 1981).

#### Classifications

#### **Clinical Classifications**

Several classifications of HCC are based on clinical and staging data (Llovet et al. 2003; review: Llovet 2007). The Barcelona Clinic Liver Cancer (BCLC) classification is an important classification for the clinical management of HCC and design of trials, endorsed by EASL and AASLD guidelines. It links stage stratification with a recommended therapy strategy. Other systems include the Cancer of the Liver Italian Program (CLIP) investigators, the Chinese University Prognostic Index (CUPI), and French staging systems, forming scoring systems predicting outcome in patients with advanced-stage disease.

### Classifications Based on Macroscopic Presentation and Growth Patterns

Virchow separated primary from secondary liver cancer. However, in the thesis of Riesenfeld performed under Virchow's guidance (1868), gallbladder cancer was still regarded as a primary liver cancer. A fundamental step into the direction of modern pathology classifications of HCCs was the work of Eggel (1901) who divided liver cancers into nodular, massive, and diffuse forms. This and subsequent classifications are discussed in detail in the paragraph on tumor macroscopy.

Originally, primary HCC was divided into two major macroscopic groups, i.e., nodular carcinoma formed of discrete nodules of varying size and massive carcinoma, consisting of one large tumor mass. Hanot and Gilbert (1888) divided HCCs into three forms, i.e., nodular cancer ("cancer nodulaire," several to numerous nodules of varying size), massive cancer ("cancer massif," one large tumor mass), and cancer with cirrhosis, i.e., they proposed a classification based on tumor morphology on the one hand and underlying liver disease on the other hand. This system had

intrinsic inconsistencies of logic and was, therefore, applicable only with difficulty or not at all, because also nodular and massive cancers of course develop in cirrhotic livers. Already Eggel (1901) put his finger on the fact that Hanot and Gilbert themselves allocated one cancer in cirrhosis each to a nodular and a massive tumor.

A major step ahead is based on the novel classification of Eggel (1901), as further outlined below. Early comments on this classification were reviewed by Wegelin (1905). Today we know that HCCs show a characteristic spectrum of macroscopic growth patterns, comprising usual and unusual gross appearances (review: Kai et al. 2012). The macroscopic growth of HCCs can be classified according to several criteria (Table 3).

 Table 3
 Classification of HCC growth patterns

Nodular I	HCC
Massive I	HCC
Diffuse H	ICC
Kaufmar	nn's classification (1958)
Large ma	ssive node
HCC in c	irrhotic liver
(a) Multip	ole nodules
(b) Diffus	se growth
(c) Comb	ination of nodules and growth
Okuda cl	assification (1984)
Expandin	g
Spreading	
Multifoca	1
Surgical	classification:
Invading	lesions ("invaders")
Expandin	g lesions ("pushers")
Peduncul	ated lesions ("hangers")
Classifica	ntion of the Liver Cancer Study Group of
Japan (L	CSGJ):
Nodular I	HCC (distinctly nodular type)
Simple	nodular (SN) type
Simple	nodular type with extranodular growth

Confluent multinodular type (CMN)

Classification according to size:

Small nodular type with indistinct margins (vaguely

(SNEG)

Nodular HCC

nodular type)

Small HCC

Solitary large HCC

In Eggel's investigations, nodular HCCs accounted for 64.6 %, massive for 23 %, and diffuse cancers for 12.4 %. Herxheimer in Germany (1930) arrived at similar figures (65 %, 28 %, and 7 %, respectively). In rare instances, a mixture of all three patterns is present. Eggel noted that mainly the massive form of HCC resulted in marked hepatomegaly, but hepatomegaly was in fact found in most of Eggel's original cases. He described a patient with a liver weight of 7 kg, but Herxheimer (1930) cites an observation by Bruzelius and Schwink reporting a liver cancer causing a liver weight of 14 kg. In some of the cases, the macroscopic presentation is dominated by liver cirrhosis, and tumor is only detectable on cut sections.

Eggel (1901) divided HCCs into two histologic groups (carcinoma solidum and carcinoma adenomatosum) and classified the macroscopic growth patterns, also based on the concepts of Hanot and Gilbert, into three patterns, i.e., nodular, massive, and diffuse. Eggel's classification of HCCs, proposed more than 100 years ago (Eggel 1901), was widely used, however, mainly for autopsy studies and less for surgical cases. The nodular type is grossly characterized by single or multiple discrete nodular neoplasms with a clear demarcation. Massive HCCs present as a large mass that almost completely replaces the left or right liver lobe. The diffuse type of HCC is characterized by a diffuse infiltration of the liver by numerous small to minute cancer nodules that macroscopically somewhat resemble cirrhotic nodules. The nodular form is the most common and is, based on data in the literature, present in 64.6 % of cases, while the massive form was found in 23 % and the diffuse form in 12.4 %. The nodular category of Eggel is sometimes applicable to surgical specimens with some difficulties only. For the adequate classification of HCC in hepatectomy specimens, the Liver Cancer Study Group of Japan proposed to divide nodular HCCs in subclasses. In 1984, Okuda and coworkers proposed a simple classification based on gross anatomical features of HCC from three disparate geographic areas, common major patterns being expanding, spreading, and multifocal (Okuda et al. 1984). Growth patterns were used as a stage

classification, e.g., the expansive-invasive-disseminative growth staging classification (Zhu et al. 2013).

normal livers were, at least in part of analyses, significantly larger in average tumor size than those occurring in cirrhotic livers (Kishi et al. 1983).

#### **Molecular Classifications**

Based on the continuously growing recognition of distinct molecular signatures found in HCC, attempts to classify HCC according to molecular features have been and are undertaken (Boyault et al. 2007; Katoh et al. 2007). Katoh and coworkers (2007) found that HCC is composed of several genetically homogeneous subclasses, each having characteristic genetic alterations. Molecular signatures are associated with distinct macroscopic growth patterns. For example, EpCAM was predominantly expressed in the confluent multinodular (CM) type of HCC which is an aggressive phenotype with poor outcome (Murakata et al. 2011). In the setting of an integrative transcriptome analysis, three robust HCC subclasses termed S1, S2, and S3 have been identified, and each of these molecular subclasses correlated with clinical parameters, including tumor size, extent of cellular differentiation, and serum AFP levels. The S1 subclass reflected aberrant activation of the WNT signaling pathway, S2 proliferation, as well as MYC and AKT activation, and S3 was associated with hepatocyte differentiation (Hoshida et al. 2009).

### Macroscopic Pathology of Main Gross Patterns of HCC

#### Size of Hepatocellular Carcinoma

The size of HCC at diagnosis varies markedly, ranging from tumors defined as early or small HCC (see the respective chapter) to very large lesions exceeding 20 cm in diameter and replacing most of the liver. Multinodular tumors with a liver weight exceeding 10 kg have been reported (Cooper et al. 1935), suggesting that, similar to other organs, HCC can grow to enormous size without leading to death through metastatic disease. HCC occurring in

#### **Nodular HCCs**

Nodular cancer HCCs forms wellin circumscribed, more or less spherical masses, mostly in a cirrhotic liver. These nodules most commonly either form simple and firm to friable masses of variable color or are closely grouped or form irregular nodular masses (Berman 1959). The color as seen on cut sections is yellowish, yellow-green, frankly green (in tumors with marked bile accumulation; "green hepatomas"), or gray to reddish. The green color of bile-rich lesions typically becomes more strong after formalin fixation. In cases where nodular cancers form groups or clusters of nodules, the color may vary from one nodule to the other, reflecting variations in differentiation (e.g., bile production) caused by progressive genomic instability or variations in regressive phenomena. Such differences can sometimes be found within one and the same nodule. Frequently, bile-storing green HCC nodules stand out against the non-bile-stained background of the cirrhotic liver. Large nodules can contains extensive hemorrhage, sometimes leaving a dark red-blue nodular lesion resembling a hemangioma (Figs. 1, 2, 3, 4, 5, 6, 7, 8, and 9). Clustered lesions form HCCs of the confluent multinodular type. In some cases, numerous nodules are present, and this presentation may be difficult to distinguish from multiple intrahepatic metastatic spread (Fig. 10). Small nodules are usually more firm than large one. The variable consistency depends on the ratio between preserved or necrotic epithelial cells versus stroma, the latter inducing firmness of the lesions when present in significant amounts. Large nodules are often soft and friable, whereby the centers are softer than the periphery owing to central necrosis. When cutting through large nodules, a creamy matter of necrotic tissue typically sticks to the blade of the knife. Macroscopically, nodular lesions tend to show an expanding growth pattern and therefore display a sharp border, but there also



**Fig. 1** Nodular hepatocellular carcinoma in a cirrhotic liver. The tumor mass displays an expanding growth pattern and shows a rather homogeneous cut surface (formalin-fixed resection specimen)



**Fig. 2** Nodular hepatocellular carcinoma with an expanding growth pattern in a non-cirrhotic liver. The heterogeneous cut surface shows central hemorrhage and few *yellow* areas with necrosis and fatty change

lesions with a blurred border due to a macroscopically invasive phenotype. Mainly in large tumors, central necrosis, sometimes with cyst formation, is often impressive.

Small HCCs with a diameter of less than about 2 cm are classified into two major types, i.e., small nodular type with indistinct margins (SN-IM, vaguely nodular type) and the distinctly nodular type (SN-DN; Kojiro and Nakashima 1999). SN-IM may, due to its blurred margins, be difficult to distinguish from surrounding cirrhotic nodules and/or nodular precursor lesions. SN-IM are histologically often well-differentiated lesions with so-called replacing growth at the periphery. The tumors can contain portal tracts in markedly reduced numbers. These portal tracts are deformed by invasive growth of HCC, the so-called stromal



**Fig. 3** Cut surface of a nodular hepatocellular carcinoma at higher magnification. This tumor is highly heterogeneous: the *greenish* part to the *left lower corner* exhibits bile accumulation, while the *whitish-pink* parts showed less differentiated neoplastic tissue. This tumor heterogeneity can cause bioptic sampling errors in regard to typing and grading. Extensive necrosis and hemorrhage are observed (*center* and *upper right corner*)



**Fig. 4** Large hepatocellular carcinoma, nodular type with multinodular interior composition. The cut surface is characterized by *brown* and *greenish* areas (bile accumulation) and several *white* necrotic foci. In this non-fixed specimen, tumor tissue bulges from the cut surface

invasion, which is a very important diagnostic element. In some studies, these lesions have been termed, early HCC (Kojiro 2007), a lesion further discussed in a separate chapter.

As seen in Table 3, Japanese investigators have subdivided nodular HCCs into three subcategories, in order to cope with imaging findings surgically resected HCCs (The Liver Cancer Study Group of Japan (LCSGJ) 2003). In this